

New South Wales

# Crimes (Sentencing Procedure) Amendment (Sentencing Options) Bill 2017

## **Explanatory note**

This explanatory note relates to this Bill as introduced into Parliament.

The following Bills are cognate with this Bill:

Justice Legislation Amendment (Committals and Guilty Pleas) Bill 2017

Crimes (High Risk Offenders) Amendment Bill 2017

### Overview of Bill

The object of this Bill is to improve the availability and nature of community-based sentencing options that are among the options for courts when sentencing offenders.

In dealing with sentencing options, the Bill:

- (a) abolishes suspended sentences, good behaviour bonds, community service orders and home detention orders, and
- (b) enhances intensive correction orders (including permitting home detention conditions to be imposed), and
- (c) creates community correction orders and conditional release orders (to replace community service orders and good behaviour bonds).

The Bill also contains provisions about sentencing domestic violence offenders, and other matters, including savings and transitional provisions and consequential amendments to other Acts.

## Outline of provisions

Clause 1 sets out the name (also called the short title) of the proposed Act.

Clause 2 provides for the commencement of the proposed Act by proclamation.

# Schedule 1 Amendment of Crimes (Sentencing Procedure) Act 1999 No 92—principal amendments

### Abolition of suspended sentences

This is achieved by the proposed repeal of section 12 by **Schedule 1** [14].

### Abolition of good behaviour bonds

This is achieved by the proposed repeal of sections 9 and 12 and the substitution of section 10 (1) (b) by **Schedule 1 [9], [14] and [10]**, respectively.

### Abolition of home detention orders

This is achieved by the proposed repeal of section 6 and Part 6 by Schedule 1 [6] and [30], respectively.

#### Intensive correction orders

An intensive correction order may be imposed by a court that has sentenced an offender to imprisonment but which enables the offender to serve the sentence in the community under strict conditions (including home detention if the court so orders).

The enhancement of intensive correction orders is achieved by the proposed substitution of section 7 and Part 5 by **Schedule 1 [7] and [29]**, respectively. Proposed section 7 authorises the making of intensive correction orders, and is subject to proposed Part 5.

Proposed Part 5 (inserted by **Schedule 1 [29]**) contains provisions relating to the making of intensive correction orders and providing for their duration and commencement, the conditions to which they are subject and other matters. These provisions include the following:

Proposed section 66 provides that community safety is the paramount consideration when a court is deciding whether to make an intensive correction order and gives guidance as to the consideration by the court of community safety and other matters.

Proposed section 67 provides that an intensive correction order cannot be made for specified offences, including murder, manslaughter, certain sexual offences, terrorism offences, and offences involving the discharge of a firearm.

Proposed section 68 provides that an intensive correction order can only be made in respect of imprisonment where the sentence of imprisonment does not exceed 2 years (or 3 years in the case of imprisonment for multiple offences).

Proposed section 69 requires the sentencing court to consider an assessment report and other evidence from a community corrections officer when considering the suitability of an offender for an intensive correction order.

Proposed section 70 provides that the term of an intensive correction order is the same as the term or terms of imprisonment in respect of which the order is made, unless the order is sooner revoked. Revocation could be effected on appeal or by the State Parole Authority following a breach of the order by the offender.

Proposed section 71 provides for the commencement of intensive correction orders.

Proposed section 72 deals generally with conditions on intensive correction orders.

Conditions are intended to be stricter than those for community correction orders and conditional release orders. Conditions are of the following kinds:

- **Standard conditions** (see proposed section 73). Standard conditions require the offender not to commit any offence and require the offender to submit to supervision.
- Additional conditions (see proposed section 73A), at least 1 of which must be imposed. Additional conditions can relate to home detention, electronic monitoring, curfew (with no specific limit on hours), community service work (not exceeding 750 hours in total), rehabilitation or treatment, abstention from alcohol or drugs, non-association with particular persons, or place restriction.

• **Further conditions** (see proposed section 73B). Further conditions may be imposed but must not be inconsistent with standard or additional conditions.

Conditions may also be imposed by the State Parole Authority under the *Crimes (Administration of Sentences) Act 1999* (see sections 81A and 164 of that Act as proposed to be inserted by **Schedule 3 [5] and [19]**, respectively).

Conditions relating to supervision, curfew, non-association and place restriction are subject to suspension under proposed section 82A of the *Crimes (Administration of Sentences) Act 1999* (see **Schedule 3 [5]**).

### **Community correction orders**

A community correction order may be made by the sentencing court as an alternative to imposing imprisonment.

The creation of community correction orders is achieved by the proposed substitution of section 8 and Part 7 by **Schedule 1 [8] and [31]**, respectively. Proposed section 8 authorises the making of community correction orders, and is subject to proposed Part 7.

Proposed Part 7 (inserted by **Schedule 1 [31]**) contains provisions relating to the duration of community correction orders and their commencement and the conditions to which they are subject. It contains sections 84–91. These provisions include the following:

Proposed section 84 states that the Part applies where a court is considering, or has made, a community correction order.

Proposed section 85 provides that the maximum term of a community correction order is 3 years. Proposed section 86 provides that a community correction order commences on the date on which it is made.

Proposed section 87 deals generally with conditions on community correction orders.

Conditions on community correction orders are intended to be less strict than those for intensive correction orders but stricter than those for conditional release orders. Conditions are of the following kinds:

- **Standard conditions** (see proposed section 88). Standard conditions require the offender not to commit any offence and require the offender to appear before the court if called on to do so.
- Additional conditions (see proposed section 89). Additional conditions can relate to curfew (not exceeding 12 hours in 24 hours), community service work (not exceeding 500 hours in total), rehabilitation or treatment, abstention from alcohol or drugs, non-association with particular persons, place restriction, or supervision. Additional conditions cannot relate to home detention, electronic monitoring or curfew.
- **Further conditions** (see proposed section 90). Further conditions may be imposed but must not be inconsistent with standard or additional conditions.

Conditions relating to supervision, curfew, non-association and place restriction are subject to suspension under proposed section 107E of the *Crimes (Administration of Sentences) Act 1999* (see **Schedule 3 [13]**).

Proposed section 91 sets out some powers in connection with applications made by the offender for the imposition, variation or revocation of additional or further conditions.

### Conditional release orders

A conditional release order may be made by a court where the offence is relatively trivial and the court has considered the offender's antecedents and other matters.

The creation of conditional release orders is achieved by the proposed substitution of section 9 and Part 8 by **Schedule 1 [9] and [31]**, respectively. Proposed section 9 authorises the making of conditional release orders, and is subject to proposed Part 8. A conditional release order may also be made for an offence expressed to be punishable only by a fine.

Proposed section 9 (1) provides that a conditional release order may be made if either:

- the court proceeds to conviction—this aspect is consistent with existing section 9 as regards good behaviour bonds imposed under that section as an alternative to imprisonment, or
- the court does not proceed to conviction but makes an order under proposed paragraph (b) of section 10 (1)—this aspect is consistent with existing section 10 (1) (b) as regards good behaviour bonds imposed under section 10.

Proposed Part 8 (inserted by **Schedule 1 [31]**) contains provisions relating to the duration of conditional release orders and their commencement and the conditions to which they are subject. It contains sections 94–100. These provisions include the following:

Proposed section 94 states that the Part applies where a court is considering, or has made, a conditional release order.

Proposed section 95 provides that the maximum term of a conditional release order is 2 years.

Proposed section 96 provides that a conditional release order commences on the date on which it is made.

Proposed section 97 deals generally with conditions on conditional release orders.

Conditions on conditional release orders are intended to be less strict than those for intensive correction orders and community correction orders. Conditions are of the following kinds:

- **Standard conditions** (see proposed section 98). Standard conditions require the offender not to commit any offence and require the offender to appear before the court if called on to do so.
- Additional conditions (see proposed section 99). Additional conditions can relate to rehabilitation or treatment, abstention from alcohol or drugs, non-association with particular persons, place restriction, or supervision. Additional conditions cannot relate to home detention, electronic monitoring, curfew or community service work.
- **Further conditions** (see proposed section 99A). Further conditions may be imposed but must not be inconsistent with standard or additional conditions.

Conditions relating to suspension, non-association and place restriction are subject to suspension under proposed section 108E of the *Crimes (Administration of Sentences) Act 1999* (see **Schedule 3 [13]**).

Proposed section 100 sets out some powers in connection with applications made by the offender for the imposition, variation or revocation of additional or further conditions.

### **Domestic violence offenders**

**Schedule 1 [4]** inserts proposed sections 4A and 4B.

Proposed section 4A provides a presumption that a court must impose a sentence of full-time detention or make a supervised order in respect of a domestic violence offender. A supervised order is an intensive correction order, community correction order or conditional release order that is subject to a condition requiring the offender to submit to supervision.

Proposed section 4B contains provisions to be observed by courts when considering:

- whether to make an intensive correction order, community correction order or conditional release order in respect of a domestic violence offender, and
- whether to impose a home detention condition on an intensive correction order in respect of a domestic violence offender.

### **Assessment reports**

Proposed Division 4B of Part 2 deals with assessment reports (see **Schedule 1 [17]**). The purpose of an assessment report is to assist a sentencing court to determine the appropriate sentence and conditions to impose on the offender during sentencing proceedings. An assessment report is to be made by a community corrections officer or juvenile justice officer.

### Multiple orders

Proposed Division 4C of Part 2 deals with circumstances where more than 1 relevant order is potentially in force at the same time in respect of an offender (see **Schedule 1 [17]**). The term *relevant orders* is defined in proposed section 17E to mean intensive correction orders, community correction orders or conditional release orders (or any combination of 1 or more of them).

Proposed section 17F (1) provides that only 1 relevant order can be in force at the same time in respect of the same offence in relation to the same offender. Subject to that, proposed section 17F (2) provides that, if 2 or more offences are involved, then 2 or more relevant orders can be in force. The section goes on to provide guidance as to which kind of order prevails over other kinds of orders.

Proposed section 17G deals with circumstances where multiple orders made in respect of an offender impose community service work conditions for hours that exceed the relevant totals contemplated by the *Crimes (Sentencing Procedure) Act 1999* as proposed to be amended (a maximum of 750 hours is contemplated for such a condition imposed on an intensive correction order, and a maximum of 500 hours is contemplated for such a condition imposed on a community correction order). The proposed section gives guidance as to how any excess is to be managed.

Proposed section 17H deals with circumstances where multiple orders made in respect of an offender impose curfew conditions. If all the orders are intensive correction orders, the proposed section does not affect the curfew conditions, as the *Crimes (Sentencing Procedure) Act 1999* as proposed to be amended does not have a specific limit on the curfew hours. If the orders are all community correction orders, or are a combination of 1 or more intensive correction orders and 1 or more community correction orders, the proposed section gives guidance as to the maximum number of curfew hours that are to be observed and as to how any excess is to be managed.

### Explanation to offender of order and notice of order to be given

Proposed section 17I (see **Schedule 1 [17]**) requires the court to take reasonable steps to explain (in language that the offender can readily understand) the offender's obligations under an intensive correction order, community correction order or conditional release order made in respect of the offender and the consequences of failing to comply with those obligations.

Proposed section 17J (see **Schedule 1 [17]**) requires notice of an intensive correction order, community correction order or conditional release order to be given to the offender and Corrective Services NSW. Notice need not be given to Corrective Services NSW unless the order is subject to a supervision condition or a community service work condition.

### Other amendments

The Bill makes other amendments to the Crimes (Sentencing Procedure) Act 1999 of a minor, ancillary or consequential nature.

# Schedule 2 Amendment of Crimes (Sentencing Procedure) Act 1999 No 92—savings and transitional provisions

Savings and transitional provisions (referred to as "clauses") are proposed to be inserted in Schedule 2 to the *Crimes (Sentencing Procedure) Act 1999*. The clauses deal with existing matters that are in force at the commencement of the proposed legislation. The following is a summary:

- (a) existing home detention orders are converted to intensive correction orders each of which is subject to a home detention condition (see proposed clause 71),
- (b) existing intensive correction orders continue in operation but with revised conditions (see proposed clause 72),
- (c) existing community service orders are converted to community correction orders (see proposed clause 73),

- (d) existing good behaviour bonds under section 9 are converted to community correction orders (see proposed clause 74),
- (e) existing good behaviour bonds under section 10 (1) (b) are converted to conditional release orders (see proposed clause 75),
- (f) existing suspended sentence orders and good behaviour bonds under section 12 continue in operation for a maximum of 3 years (see proposed clause 76),
- (g) special provision is made for any existing periodic detention orders under former legislation that are still in force (see proposed clauses 79–82).

## Schedule 3 Amendment of Crimes (Administration of Sentences) Act 1999 No 93

### Administration of intensive correction orders

Provisions relating to intensive correction orders are currently set out in Part 3 of the *Crimes (Administration of Sentences) Act 1999*, which consists of sections 80–93. Schedule 3 proposes amendments to Part 3.

Section 81 is proposed to be substituted by **Schedule 3** [5], which states that conditions are imposed under the *Crimes (Sentencing Procedure) Act 1999* or by the State Parole Authority under the *Crimes (Administration of Sentences) Act 1999*.

Proposed section 81A is inserted by **Schedule 3 [5]** and authorises the State Parole Authority to impose conditions on intensive correction orders. These conditions are additional to those imposed by courts at the time of sentence.

Section 82 is proposed to be substituted by **Schedule 3 [5]**, which provides that the administration of intensive correction orders is to be as prescribed by the regulations, and that the obligations of an offender under an intensive correction order, including a condition of an order, are to be as so prescribed.

Proposed section 82A is inserted by **Schedule 3 [5]** and provides that a condition of an intensive correction order relating to supervision, curfew, non-association and place restriction is subject to suspension (either for a specified period or indefinitely, and either conditionally or unconditionally).

Section 83 is proposed to be substituted by **Schedule 3 [6]**, which provides that an intensive correction order expires at the end of the sentence or sentences to which it relates, unless it is sooner revoked.

Provisions relating to breaches of intensive correction orders are contained in proposed substitutions of sections 163 and 164 (see **Schedule 3 [19]**). Proposed section 163 deals with actions that may be taken by a community corrections officer for a breach, and proposed section 164 deals with actions that may be taken by the State Parole Authority for a breach.

Proposed sections 163 and 164 provide a hierarchy of actions that can be taken in the event of a failure by an offender to comply with an intensive correction order. Under these provisions:

- (a) A community corrections officer may record the non-compliance with no further action, give an informal warning, give or arrange for a formal warning, give a reasonable direction about the non-compliant behaviour or impose a curfew.
- (b) If the failure to comply is more serious, the Commissioner or a community corrections officer may refer the matter to the State Parole Authority.
- (c) The State Parole Authority may deal with a breach of an intensive correction order by recording the breach with no further action, giving a formal warning, changing the conditions of the order or revoking the order. Express authority is conferred to impose a condition of home detention, a requirement to submit to the use of an electronic monitoring device and other ancillary related conditions.

### Administration of community correction orders

Proposed Part 4B, which deals with the administration of community correction orders, is inserted by **Schedule 3** [13]. It contains sections 107A–107E.

Proposed section 107A states that the conditions of a community correction order are imposed under the *Crimes (Sentencing Procedure) Act 1999* or under section 107D.

Proposed section 107B provides that the administration of community correction orders is to be as prescribed by the regulations, and that the obligations of an offender under a community correction order, including a condition of an order, are to be as so prescribed.

Proposed section 107C deals with a breach of a community correction order. A breach can be dealt with by the court that made the order, any other court of like jurisdiction or (with the offender's consent) any other court of superior jurisdiction. The court may decide to take no further action, vary the condition of the order, impose new conditions on the order or revoke the order.

Proposed section 107D provides that if a court revokes a community correction order, the court may re-sentence the offender.

Proposed section 107E provides that a condition of a community correction order relating to supervision, curfew, non-association and place restriction is subject to suspension (either for a specified period or indefinitely, and either conditionally or unconditionally).

### Administration of conditional release orders

Proposed Part 4C, which deals with the administration of conditional release orders, is inserted by **Schedule 3 [13]**. It contains sections 108A–108E.

Proposed section 108A states that the conditions of a conditional release order are imposed under the *Crimes (Sentencing Procedure) Act 1999* or under section 108D.

Proposed section 108B provides that the administration of conditional release orders is to be as prescribed by the regulations, and that the obligations of an offender under a conditional release order, including a condition of an order, are to be as so prescribed.

Proposed section 108C deals with a breach of a conditional release order. A breach can be dealt with by the court that made the order, any other court of like jurisdiction or (with the offender's consent) any other court of superior jurisdiction. The court may decide to take no further action, vary the condition of the order, impose new conditions on the order or revoke the order.

Proposed section 108D provides that if a court revokes a conditional release order, the court may re-sentence the offender.

Proposed section 108E provides that a conditional release order relating to supervision, curfew, non-association and place restriction is subject to suspension (either for a specified period or indefinitely, and either conditionally or unconditionally).

### Omission of provisions relating to superseded orders and bonds

Part 4, which currently relates to imprisonment by way of home detention under home detention orders, is repealed by **Schedule 3** [12]. Other provisions relating to, or references to, home detention orders, community service orders and good behaviour bonds are deleted by Schedule 3.

### **ICO Management Committee**

Provisions relating to the establishment and functions of the Intensive Correction Orders Management Committee are repealed (see Schedule 3 [10] and [45]).

### Other amendments

The Bill makes other amendments to the Crimes (Administration of Sentences) Act 1999 of a minor, consequential or transitional nature.

### Schedule 4 Amendment of other legislation

A number of other Acts and statutory instruments are amended, to make consequential amendments arising from amendments made by Schedules 1–3, arising in particular from:

- (a) the proposed abolition of home detention orders, community service orders and good behaviour bonds, and
- (b) the enhancement of intensive correction orders, and
- (c) the creation of community correction orders and conditional release orders.

The following are proposed to be amended:

Anzac Memorial (Building) Act 1923 No 27

Bail Act 2013 No 26

Births, Deaths and Marriages Registration Act 1995 No 62

Child Protection (Offenders Registration) Act 2000 No 42

Child Protection (Offenders Registration) Regulation 2015

Child Protection (Working with Children) Act 2012 No 51

Children (Criminal Proceedings) Act 1987 No 55

Civil Liability Act 2002 No 22

Crimes Act 1900 No 40

Crimes (Appeal and Review) Act 2001 No 120

Crimes (Forensic Procedures) Act 2000 No 59

Crimes (Interstate Transfer of Community Based Sentences) Act 2004 No 72

Criminal Appeal Act 1912 No 16

Criminal Procedure Act 1986 No 209

Criminal Records Act 1991 No 8

Dangerous Goods (Road and Rail Transport) Act 2008 No 95

Drug Court Act 1998 No 150

Education Act 1990 No 8

Fines Act 1996 No 99

Fisheries Management Act 1994 No 38

Graffiti Control Act 2008 No 100

Jury Act 1977 No 18

Motor Dealers and Repairers Regulation 2014

Privacy Code of Practice (General) 2003

Security Industry Regulation 2016

Summary Offences Act 1988 No 25

Supreme Court Act 1970 No 52

Tow Truck Industry Regulation 2008



## New South Wales

# Crimes (Sentencing Procedure) Amendment (Sentencing Options) Bill 2017

## **Contents**

			Page
	1	Name of Act	2
	2	Commencement	2
Schedule 1		Amendment of Crimes (Sentencing Procedure) Act 1999 No 92—principal amendments	3
Schedule 2		Amendment of Crimes (Sentencing Procedure) Act 1999 No 92—savings and transitional provisions	21
Schedule 3		Amendment of Crimes (Administration of Sentences) Act 1999 No 93	29
Schedule 4		Amendment of other legislation	42



### New South Wales

# Crimes (Sentencing Procedure) Amendment (Sentencing Options) Bill 2017

No , 2017

### A Bill for

An Act to amend the *Crimes (Sentencing Procedure) Act 1999* with respect to sentencing options available to courts for persons found guilty of offences; to abolish home detention orders, community service orders, suspended sentences and good behaviour bonds; to make consequential and other amendments to the *Crimes (Administration of Sentences) Act 1999* and other legislation; and for other purposes.

See also the Justice Legislation Amendment (Committals and Guilty Pleas) Bill 2017 and the Crimes (High Risk Offenders) Amendment Bill 2017.

Legislature of New South Wales enacts:	1
Name of Act	2
This Act is the Crimes (Sentencing Procedure) Amendment (Sentencing Options) Act 2017.	3
Commencement	5
This Act commences on a day or days to be appointed by proclamation.	6
	Name of Act  This Act is the Crimes (Sentencing Procedure) Amendment (Sentencing Options)  Act 2017.  Commencement

Scł	nedu	le 1	Amendment of Crimes (Sentencing Procedure) Act 1999 No 92—principal amendments	1 2				
[1]	Sect	ion 3	Interpretation	3				
	Insert in alphabetical order in section 3 (1):							
	Commissioner means the Commissioner of Corrective Services, Department of Justice.							
	comi	nunity	correction order means an order referred to in section 8.	6				
			corrections officer has the same meaning as it has in the Crimes atton of Sentences) Act 1999.	7 8				
		•	service work condition—see sections 73A and 89.	g				
			<i>I release order</i> means an order referred to in section 9.	10				
			Services NSW has the same meaning as it has in the Crimes (Administration of Act 1999.	11 12				
			<i>iolence offence</i> has the same meaning as it has in the <i>Crimes (Domestic and Violence) Act 2007</i> .	13 14				
	home	e deter	ntion condition—see section 73A.	15				
	<b>juver</b> Act l		stice officer has the same meaning as it has in the Children (Detention Centres)	16 17				
			<b>Thority</b> means the State Parole Authority constituted by section 183 of the <i>Crimes</i> action of Sentences) Act 1999.	18 19				
	proceed to a conviction includes record a conviction.							
	supe	rvisioi	a condition—see sections 73, 89 and 99.	21				
[2]	Sect hom	ion 3 ( e dete	(1), definitions of "community service order", "good behaviour bond" and ention order"	22 23				
	Omit	the de	efinitions.	24				
[3]	Section 3 (1), definition of "full-time detention"							
	Omit	Omit the definition. Insert instead:						
	full-time detention means detention in a correctional centre.							
[4]	Sect	ions 4	A and 4B	28				
	Insert after section 4:							
	4A		nestic violence offenders—requirement for full-time detention or ervision	30 31				
		(1)	If a court finds a person guilty of a domestic violence offence, the court must impose on the person either:	32 33				
			(a) a sentence of full-time detention, or	34				
			(b) a supervised order.	35				
		(2)	However, the court is not required to impose either of those sentencing options if the court is satisfied that a different sentencing option is more appropriate in the circumstances and gives reasons for reaching that view.	36 37 38				
		(3)	For the purposes of this section, a <i>supervised order</i> is an order (being an intensive correction order, community correction order or conditional release order) that is subject to a supervision condition.	39 40 41				

	4B	Dom	estic violence offenders—protection and safety of victims	1		
		(1)	An intensive correction order must not be made in respect of:	2		
			(a) a sentence of imprisonment for a domestic violence offence, or	3		
			(b) an aggregate sentence of imprisonment for 2 or more offences, any 1 or more of which is a domestic violence offence,	4 5		
			unless the sentencing court is satisfied that the victim of the domestic violence offence, and any person with whom the offender is likely to reside, will be adequately protected (whether by conditions of the intensive correction order or for some other reason).	6 7 8		
		(2)	If the sentencing court finds a person guilty of a domestic violence offence, the court must not impose a home detention condition if the court reasonably believes that the offender will reside with the victim of the domestic violence offence.	10 11 12 13		
		(3)	Before making a community correction order or conditional release order in respect of a person whom the sentencing court finds guilty of a domestic violence offence, the court must consider the safety of the victim of the offence.	14 15 16 17		
[5]	Sect	ion 5 F	Penalties of imprisonment	18		
	Omit	or ho	ome detention order" from section 5 (5).	19		
[6]	Sect	ion 6 H	Home detention	20		
	Omit	the se	ection.	21		
[7]	Section 7					
	Omit the section. Insert instead:					
	7	Inten	nsive correction orders	24		
		(1)	A court that has sentenced an offender to imprisonment in respect of 1 or more offences may make an intensive correction order directing that the sentence or sentences be served by way of intensive correction in the community.	25 26 27		
		(2)	If the court makes an intensive correction order directing that a sentence of imprisonment be served by way of intensive correction in the community, the court is not to set a non-parole period for the sentence.	28 29 30		
		(3)	This section does not apply to an offender who is under the age of 18 years.	31		
		(4)	This section is subject to the provisions of Part 5.	32		
			<b>Note.</b> Among other matters, Part 5 provides that a single offence cannot be the subject of an intensive correction order if the imprisonment imposed exceeds 2 years, and that multiple offences cannot be the subject of an intensive correction order or orders if the imprisonment imposed exceeds 3 years.	33 34 35 36		
[8]	Section 8					
	Omit	the se	ection. Insert instead:	38		
	8	Com	munity correction orders	39		
		(1)	Instead of imposing a sentence of imprisonment on an offender, a court that has convicted a person of an offence may make a community correction order in relation to the offender.	40 41 42		

	(2)	corre	ommunity service work condition must not be imposed on a community ection order made in relation to an offender to whom the <i>Children mmunity Service Orders</i> ) <i>Act 1987</i> applies.	2 ;		
	(3)	This	section is subject to the provisions of Part 7.	4		
[9]	Section 9			ţ		
	Omit the se	ection.	Insert instead:	(		
	9 Con	litional release orders				
	(1)	offer	Instead of imposing a sentence of imprisonment or a fine (or both) on an offender, a court that finds a person guilty of an offence may make a conditional release order discharging the offender, if:			
		(a)	the court proceeds to conviction, or	1		
		(b)	the court does not proceed to conviction but makes an order under section 10 (1) (b).	12 13		
	(2)		eciding whether to make a conditional release order with a conviction, the encing court is to have regard to the following factors:	14 15		
		(a)	the person's character, antecedents, age, health and mental condition,	16		
		(b)	whether the offence is of a trivial nature,	17		
		(c)	the extenuating circumstances in which the offence was committed,	18		
		(d)	any other matter that the court thinks proper to consider.	19		
		Note section	e. These factors are considered under section 10 in respect of an order under on 10 (1) (b) in connection with a conditional release order without a conviction.	20 2		
	(3)	To a	void doubt and without limitation:	22		
		(a)	a fine and a conditional release order cannot be imposed in relation to the offender in respect of the same offence, and	23 24		
		(b)	a conditional release order with a conviction may be made as an alternative to imposing a fine.	25 26		
	(4)	This	section is subject to the provisions of Part 8.	27		
[10]	Section 10	Dism	issal of charges and conditional discharge of offender	28		
	Omit section	n 10 (	(1) (b). Insert instead:	29		
		(b)	an order discharging the person under a conditional release order (in which case the court proceeds to make a conditional release order under section 9),	30 32		
[11]	Section 10	(1A)		33		
	Insert after	section	n 10 (1):	34		
	(1A)	secti	eference in any legislation (including this Act) to an order under this ion includes, in the case of an order under subsection (1) (b), a reference conditional release order made under section 9 pursuant to that paragraph.	35 36 37		
[12]	Section 10	(2) (b	o)	38		
	Omit the pa	aragrap	ph. Insert instead:	39		
	•	(b)	that it is expedient to discharge the person under a conditional release order.	4( 4 <sup>-</sup>		

[13]	Sect	ion 10	(4)		1
	Inser	t at the	end o	of the subsection:	2
				Certain other Acts and regulations contain provisions to the effect that an order	3
			certai	r this section made in respect of an offence is to be treated as a conviction for in purposes of the legislation concerned. Accordingly, those provisions apply to an	4 5
				under subsection (1) (b) in respect of the offence and a conditional release order pursuant to that paragraph.	6 7
[14]	Sect	ion 12	Susp	ended sentences	8
	Omi	t the se	ction.		9
[15]		ion 13 alties o		munity service orders and good behaviour bonds to be alternative	10 11
	Omi	t the se	ction.		12
[16]	Sect	ion 14	Fines	as an additional penalty to good behaviour bond	13
	Omi	t the se	ction.		14
[17]	Part	2, Divi	sions	4B and 4C	15
	Inser	t after	section	n 17A:	16
	Divi	sion	4B	Assessment reports	17
	17B	Defir	ition	of and provisions relating to "assessment report"	18
		(1)	In th	is Division:	19
				ssment report means a report made by a community corrections officer or renile justice officer under this Part.	20 21
		(2)	the a	purpose of an assessment report is to assist a sentencing court to determine appropriate sentence options and conditions to impose on the offender ag sentencing proceedings.	22 23 24
		(3)		ssessment report is made by a community corrections officer or a juvenile ce officer.	25 26
		(4)		regulations may make provision for or with respect to matters to be essed in, and the preparation and furnishing of, an assessment report.	27 28
	17C	Requ	iest fo	or assessment report	29
		(1)	Exce	ept as provided by section 17D:	30
			(a)	the sentencing court may request, but is not obliged to request, an assessment report on an offender, and	31 32
			(b)	the request may be made at any time during the sentencing proceedings, whether before or after imposing a sentence.	33 34
		(2)	sente	court refers an offender for assessment in relation to a sentence and a ence of imprisonment has been imposed in respect of the offence erned:	35 36 37
			(a)	the referral stays the execution of the sentence and the operation of section 48, and	38 39
			(b)	the offender is to be remanded in custody, or granted bail in accordance with the <i>Bail Act 2013</i> ,	40 41
			until	the court decides whether or not to make an intensive correction order.	42

17D	Requ	uiremer	nt for assessment report	•		
	(1)	an off	entencing court must not make an intensive correction order in respect of ender unless it has obtained a relevant assessment report in relation to fender.	2		
	(2)	intens	entencing court must not impose a home detention condition on an ive correction order unless it has obtained an assessment report relating imposition of such a condition in relation to the offender.			
	(3)	impos unless	entencing court must not request an assessment report relating to the ition of a home detention condition on an intensive correction order it has imposed a sentence of imprisonment on the offender for a fied term.	8 9 10 11		
	(4)	an int obtain	entencing court must not impose a community service work condition on ensive correction order or community correction order unless it has sed an assessment report relating to the imposition of such a condition in on to the offender.	12 13 14 15		
	(5)		ssessment reports referred to in this section may be in the 1 report or in than 1 report.	16 17		
		Note. S condition	See also sections 73A (3) and 89 (4) regarding the imposition of home detention ons and community service work conditions.	18 19		
Divi	sion	4C	Provisions relating to certain orders	20		
17E	Definitions					
		In this	Division:	22		
		releva of the	<i>int orders</i> means the following orders (or any combination of 1 or more m):	23 24		
		(a)	intensive correction orders,	25		
		(b)	community correction orders,	26		
		(c)	conditional release orders.	27		
17F	Mult	iple ord	lers	28		
	(1)	Only offence	1 relevant order can be in force at the same time in respect of the same see in relation to the same offender.	29 30		
	(2)		et to subsection (1), 2 or more relevant orders can be in force at the same in respect of 2 or more offences in relation to the same offender.	3 <sup>2</sup>		
	(3)	a com	e purposes of subsection (1), an intensive correction order prevails over munity correction order, and a community correction order prevails over litional release order.	33 34 38		
	(4)	there	the purposes of subsection (2) and subject to sections 17G and 17H, if is an inconsistency as to how any conditions of the relevant orders the together, then to the extent of the inconsistency:	36 37 38		
			a condition of an intensive correction order prevails over a condition of a community correction order, and	39 40		
			a condition of a community correction order prevails over a condition of a conditional release order, and	4° 42		
		(c)	despite paragraphs (a) and (b), a standard condition prevails over a condition that is not a standard condition.	43 44		

7G		munity service work conditions under multiple orders
	(1)	A relevant order (the <i>new order</i> ) may not be made if the sum of:
		(a) the number of hours of community service work to be performed under the new order, and
		(b) the number of hours of community service work remaining to be performed under any other relevant order (an <i>existing order</i> ),
		exceeds 750 hours (if any one of the orders is an intensive correction order) or 500 hours (if all the orders are community correction orders).
		<b>Note.</b> Community service work conditions can be imposed on intensive correction orders and community correction orders, but cannot be imposed on conditional release orders.
	(2)	In calculating the sum referred to in subsection (1), the hours of community service work to be performed under the new order are to be disregarded to the extent to which they run concurrently with those to be performed under any existing order.
	(3)	The hours of community service work to be performed under the new order are taken to run concurrently with those to be performed under any existing order.
7H	Curf	ew conditions under multiple orders
	(1)	This section applies where 2 or more curfew conditions apply under 2 or more relevant orders in respect of the same period of 24 hours (the <i>period of 24 hours</i> ).
		<b>Note.</b> Curfew conditions can be imposed on intensive correction orders and community correction orders, but cannot be imposed on conditional release orders.
	(2)	If all the relevant orders are intensive correction orders, this section does not affect the curfew conditions.
	(3)	If all the relevant orders are community correction orders, the following provisions apply:
		(a) The offender cannot be required to observe a curfew in respect of more than 12 hours in the period of 24 hours. Any excess is to be disregarded.
		(b) The offender is required in the period of 24 hours to observe only the curfew imposed by the 1 curfew condition that specifies more hours than the other or others.
	(4)	If at least 1 of the relevant orders is an intensive correction order and at least 1 is a community correction order, the following provisions apply:
		(a) This section does not affect any curfew condition imposed on an intensive correction order.
		(b) The offender cannot be required, as a result of the curfew conditions imposed on the relevant orders, to observe a curfew in respect of more than the greater of:
		(i) the hours required by curfew conditions imposed on the intensive correction order or intensive correction orders in the period of 24 hours, or
		(ii) 12 hours in the period of 24 hours.
		Any excess is to be disregarded.
		(c) In determining the number of hours under 2 or more curfew conditions imposed on 2 or more community correction orders, regard is to be had only to the 1 curfew condition that specifies more hours than the other

or others.

		(5)	make hours	regulations under the <i>Crimes (Administration of Sentences) Act 1999</i> may e provision for or with respect to the manner of determining numbers of s for the purposes of this section and any excess to be disregarded under section.	2		
	17I	Explanation of relevant order to offender					
		(1)	must	ing made a relevant order in relation to an offender, the sentencing court tensure that reasonable steps are taken to explain to the offender (in the uage that the offender can readily understand):			
			(a)	the offender's obligations under the order, and	Ş		
			(b)	the consequences that may follow if the offender fails to comply with those obligations.	10 17		
		(2)	A rel	levant order is not invalidated by a failure to comply with this section.	12		
	17J	Noti	ce of r	elevant order to be given	13		
		(1)	offic	oon as practicable after a relevant order is made, the registrar or another er of the sentencing court must cause notice of the order to be given to the order and to Corrective Services NSW.	14 15 16		
		(2)		notice must include such information about the relevant order as may be cribed by the regulations.	17 18		
		(3)	unles	ce of an order does not need to be given to Corrective Services NSW ss the order is subject to a supervision condition or a community service condition.	19 20 21		
		(4)	A rel	levant order is not invalidated by a failure to comply with this section.	22		
[18]	Sect	ion 24	Court	t to take other matters into account	23		
	Omit	"com	munity	y service order, good behaviour bond" from section 24 (b).	24		
	Inser	t inste	ad "co	mmunity correction order, conditional release order".	2		
[19]	Sect	ion 24	(b) (i)	and (ii)	26		
	Omit	or bo	ond" w	herever occurring.	27		
[20]	Sect	ion 24	(c)		28		
	Omit	or go	ood be	haviour bond" wherever occurring.	29		
[21]	Sect	ion 25	Local	Court not to impose certain penalties if offender is absent	30		
	Omit	section	on 25 (	1) (c)–(e). Insert instead:	31		
			(d)	a community correction order,	32		
			(e)	a conditional release order,	33		
[22]	Sect	ion 31	Defin	itions	34		
				etention order or community service order" from paragraph (b) of the ose a penalty.	35 36		
	Inser	t inste	ad "or	community correction order".	37		

[23]	Section 43	Court may reopen proceedings to correct sentencing errors	1
		ne detention order or community service order" from paragraph (b) of the of <i>impose a penalty</i> in section 43 (6).	2
	Insert inste	ad "community correction order or conditional release order".	4
[24]	Section 43	(6), paragraph (c) of definition of "impose a penalty"	5
	Omit the pa	aragraph.	6
[25]	Section 43	6 (6), paragraph (d) of definition of "impose a penalty"	7
	Omit "10,	11 or 12". Insert instead "10 or 11".	8
[26]	Section 47	Commencement of sentence	9
	Omit section	on 47 (1). Insert instead:	10
	(1)	A sentence of imprisonment commences, subject to section 71 and to any direction under subsection (2), on the day on which the sentence is imposed.	11 12
[27]	Section 62	2 Warrant of commitment	13
	Omit section	on 62 (4). Insert instead:	14
	(4)	This section does not apply:	15
		(a) while action is being taken under Part 5 in relation to the making of an intensive correction order, or	16 17
		(b) to a sentence of imprisonment that is the subject of an intensive correction order.	18 19
[28]	Section 63	Offenders to be photographed and fingerprinted	20
	Omit "or h	ome detention order" from section 63 (2).	21
[29]	Part 5		22
	Omit the P	art. Insert instead:	23
	Part 5	Sentencing procedures for intensive correction orders	24 25
	Division	1 Preliminary	26
	64 App	lication	27
		This Part applies in circumstances in which a court is considering, or has made, an intensive correction order.	28 29
	65 Defi	nition	30
		In this Part:	31
		assessment report, in relation an offender, means an assessment report within the meaning of Division 4B of Part 2 made in respect of the offender.	32 33

Division 2			Restrictions on power to make intensive correction orders				
66	Com	munit	y safe	safety and other considerations			
	(1)	court		y safety must be the paramount consideration when the sentencing ciding whether to make an intensive correction order in relation to r.	4 5 6		
	(2)	maki	ng the	sidering community safety, the sentencing court is to assess whether order or serving the sentence by way of full-time detention is more ldress the offender's risk of reoffending.	7 8 9		
	(3)	court	must any re	ding whether to make an intensive correction order, the sentencing also consider the provisions of section 3A (Purposes of sentencing) elevant common law sentencing principles, and may consider any that the court thinks relevant.	10 11 12 13		
67	Inter	nsive c	orrec	tion order not available for certain offences	14		
	(1)			ve correction order must not be made in respect of a sentence of ent for any of the following offences:	15 16		
		(a)	mure	der or manslaughter,	17		
		(b)	a pre	escribed sexual offence,	18		
		(c)		rorism offence within the meaning of the <i>Crimes Act 1914</i> of the monwealth or an offence under section 310J of the <i>Crimes Act 0</i> ,	19 20 21		
		(d)	orde	offence relating to a contravention of a serious crime prevention or under section 8 of the <i>Crimes (Serious Crime Prevention Orders)</i> 2016,	22 23 24		
		(e)	secti	offence relating to a contravention of a public safety order under ion 87ZA of the <i>Law Enforcement (Powers and Responsibilities)</i> 2002,	25 26 27		
		(f)	an o	ffence involving the discharge of a firearm,	28		
		(g)		ffence that includes the commission of, or an intention to commit, ffence referred to in paragraphs (a)–(f),	29 30		
		(h)		ffence of attempting, or of conspiracy or incitement, to commit an nee referred to in paragraphs (a)–(g).	31 32		
	(2)		_	poses of this section:	33		
		Commonwealth Criminal Code means the Criminal Code set out in the Schedule to the Criminal Code Act 1995 of the Commonwealth.					
		firearm means a firearm as defined in the Firearms Act 1996.					
		_		sexual offence means:	37		
		(a)	bein		38 39		
			(i)	an offence the victim of which is a person under the age of 16 years, or	40 41		
			(ii)	an offence the victim of which is a person of any age and the elements of which include sexual intercourse (as defined by section 61H of that Act), or	42 43 44		
		(b)	an o	ffence under section 91D, 91E, 91F, 91G or 91H of the <i>Crimes Act</i> 0, or	45 46		

		(c)	an offence under section 91J, 91K or 91L of the <i>Crimes Act 1900</i> , being an offence the victim of which is a person under the age of 16 years, or	1 2
		(d)	an offence against section 50BA, 50BB, 50BC, 50BD, 50DA or 50DB of the <i>Crimes Act 1914</i> of the Commonwealth, being an offence the victim of which was a person under the age of 16 years, or	3 4 5
		(e)	an offence against section 71.8, 71.12, 271.4, 271.7, 272.8 (1) or (2), 272.9 (1) or (2), 272.10 (1), 272.11 (1), 272.12 (1) or (2), 272.13 (1) or (2), 272.14 (1), 272.15 (1), 272.18 (1), 272.19 (1), 272.20 (1) or (2), 273.5, 273.6, 273.7, 471.16 (1) or (2), 471.17 (1), 471.19 (1) or (2), 471.20 (1), 471.22 (1), 471.24, 471.25, 471.26, 474.19 (1), 474.20 (1), 474.22 (1), 474.23 (1), 474.24A (1), 474.25A (1) or (2), 474.25B (1), 474.26, 474.27 (1), (2) or (3), 474.27A of the Commonwealth Criminal Code, being an offence the victim of which was a person under the age of 16 years, or	6 7 8 9 10 11 12 13
		(f)	an offence against section 233BAB of the <i>Customs Act 1901</i> of the Commonwealth involving items of child pornography or child abuse material, or	15 16 17
		(g)	an offence that, at the time it was committed, was a prescribed sexual offence within the meaning of this definition.	18 19
	(3)	more	void doubt, subsection (1) extends to a sentence of imprisonment for 2 or e offences any 1 of which includes an offence referred to in that ection.	20 21 22
68	Inter	nsive c	correction orders not available where imprisonment exceeds limits	23
	(1)		ntensive correction order must not be made in respect of a single offence e duration of the term of imprisonment imposed for the offence exceeds ars.	24 25 26
	(2)	of in	ntensive correction order may be made in respect of an aggregate sentence apprisonment. However, the order must not be made if the duration of the of the aggregate exceeds 3 years.	27 28 29
	(3)		or more intensive correction orders may be made in respect of each of more offences. However, the orders must not be made if:	30 31
		(a)	the duration of the term of any individual term of imprisonment exceeds 2 years, and	32 33
		(b)	the duration of the term of imprisonment imposed for all the offences exceeds 3 years.	34 35
69	Asse	essme	nt of suitability of offender for intensive correction order	36
	(1)		eciding whether or not to make an intensive correction order, the encing court is to have regard to:	37 38
		(a)	the contents of the assessment report referred to in section 17D (1) relating to the offender, and	39 40
		(b)	such evidence from a community corrections officer as the court considers necessary for the purpose of deciding whether to make such an order.	41 42 43
	(2)	Subjo repor	ect to section 73A (3), the sentencing court is not bound by the assessment rt.	44 45

Divi	Division 3		Term and commencement	1
70	Term	of in	tensive correction order	2
		the to	ess sooner revoked, the term of an intensive correction order is the same as erm or terms of imprisonment in respect of which the order is made.  Section 68 contains provisions regarding the term or terms of imprisonment in ect of which intensive correction orders may be made.	3 4 5 6
71	Com	menc	ement of intensive correction order	7
	(1)	An ii	ntensive correction order commences on the date on which it is made.	8
	(2)	to a conc	section (1) does not apply to an intensive correction order made in relation sentence of imprisonment that is to be served consecutively (or partly urrently and partly consecutively) with some other sentence of isonment the subject of an intensive correction order.	9 10 11 12
Divi	sion	4	Conditions	13
72	Cond	ditions	s generally	14
		An ii	ntensive correction order is subject to the following conditions:	15
		(a)	standard conditions imposed by the sentencing court under section 73,	16
		(b)	any additional conditions imposed by the sentencing court under section 73A,	17 18
		(c)	any further conditions imposed by the sentencing court under section 73B,	19 20
		(d)	any conditions imposed by the Parole Authority under section 81A or 164 of the <i>Crimes (Administration of Sentences) Act 1999</i> .	21 22
		order	1. Provisions relating to a breach of obligations under an intensive correction are contained in sections 163 and 164 of the <i>Crimes (Administration of ences) Act 1999</i> .	23 24 25
		intens	<b>2.</b> After an offender is sentenced by the sentencing court, conditions of an sive correction order are imposed, varied or revoked by the Parole Authority rather the court.	26 27 28
73	Stan	dard o	conditions	29
	(1)		sentencing court must at the time of sentence impose on an intensive ection order the standard conditions of an intensive correction order.	30 31
	(2)	The	standard conditions of an intensive correction order are the following:	32
		(a)	a condition that the offender must not commit any offence,	33
		(b)	a condition that the offender must submit to supervision by a community corrections officer.	34 35
		1999	. Regulations may be made under the <i>Crimes (Administration of Sentences) Act</i> to prescribe the offender's obligations under a standard condition of an intensive ction order.	36 37 38
73A	Addi	tional	conditions	39
	(1)	sente	Idition to the standard conditions, the sentencing court must at the time of ence impose on an intensive correction order at least 1 of the additional litions referred to in subsection (2).	40 41 42
	(2)		additional conditions of an intensive correction order that are available to nposed are the following conditions (as directed by the sentencing court):	43 44
		(a)	a home detention condition,	45

		(b)	an electronic monitoring condition,	1		
		(c)	a curfew condition imposing a specified curfew,	2		
		(d)	a community service work condition requiring the performance of community service work for a specified number of hours (not exceeding 750 hours),	3 2 5		
		(e)	a rehabilitation or treatment condition requiring the offender to participate in a rehabilitation program or to receive treatment,	6		
		(f)	an abstention condition requiring abstention from alcohol or drugs or both,	8		
		(g)	a non-association condition prohibiting association with particular persons,	10 11		
		(h)	a place restriction condition prohibiting the frequenting of or visits to a particular place or area.	12 13		
	(3)	comn	sentencing court must not impose a home detention condition or nunity service work condition on an intensive correction order unless an sment report states that the offender is suitable to be the subject of such addition.	14 15 16 17		
		Note. report	The provisions of section 17D (2), (3) and (4) refer specifically to assessment to regarding home detention conditions and community service work conditions.	18 19		
	(4)		sentencing court may limit the period during which an additional ition imposed by it on an intensive correction order is in force.	20 21		
		1999	Regulations may be made under the <i>Crimes (Administration of Sentences) Act</i> to prescribe the offender's obligations under an additional condition of an sive correction order.	22 23 24		
73B	Furth	ner coi	nditions	25		
	(1)		sentencing court may at the time of sentence impose further conditions on tensive correction order.	26 27		
	(2)	This condi	section does not permit the sentencing court to impose any further itions so as to be inconsistent with:	28 29		
		(a)	any of the standard conditions of an intensive correction order, or	30		
		(b)	any of the additional conditions (whether or not imposed on the intensive correction order) referred to in section 73A (2).	31 32		
	(3)		sentencing court may limit the period during which a further condition sed by it on an intensive correction order is in force.	33 34		
Part	6 Sent	encin	g procedures for home detention orders	35		
Omit	t the Pa	ırt.		36		

[30]

[31]	Parts 7 and 8 Omit the Parts. Insert instead:					
	Par		Sentencing procedures for community correction		2	
			ord	lers	4	
	Divi	ision	1	Preliminary	5	
	84	Appl	icatio	n	6	
				Part applies in circumstances in which a court is considering, or has e, a community correction order.	7 8	
	Divi	ision	2	Term and commencement	9	
	85	Term	of co	ommunity correction order	10	
		(1)	The	term of a community correction order is the period specified in the order.	11	
		(2)	The	maximum term of a community correction order is 3 years.	12	
	86	Com	menc	ement of community correction order	13	
				mmunity correction order commences on the date on which it is made.	14	
	Divi	ision	3	Conditions	15	
	87	Cond	litions	s generally	16	
			A co	ommunity correction order is subject to the following conditions:	17	
			(a)	standard conditions imposed by the sentencing court under section 88,	18	
			(b)	any additional conditions imposed by the sentencing court under section 89,	19 20	
			(c)	any further conditions imposed by the sentencing court under section 90.	21 22	
	88	Stan	dard o	conditions	23	
		(1)		sentencing court must at the time of sentence impose on a community ection order the standard conditions of a community correction order.	24 25	
		(2)	The	standard conditions of a community correction order are the following:	26	
			(a)	a condition that the offender must not commit any offence,	27	
			(b)	a condition that the offender must appear before the court if called on to do so at any time during the term of the community correction order.	28 29	
			1999	. Regulations may be made under the <i>Crimes (Administration of Sentences) Act</i> to prescribe the offender's obligations under a standard condition of a community ction order.	30 31 32	
	89	Addi	tional	conditions	33	
		(1)		sentencing court may at the time of sentence, or subsequently on the ication of a community corrections officer or juvenile justice officer or the ider:	34 35 36	
			(a)	impose on a community correction order any of the additional conditions of a community correction order, or	37 38	

	(b)	vary	or revoke any such additional conditions imposed by it on a munity correction order.	1 2
(2)		e impo	onal conditions of a community correction order that are available osed are the following conditions (as directed by the sentencing	3 4 5
	(a)		rfew condition imposing a specified curfew (not exceeding 12 hours my period of 24 hours),	6 7
	(b)	com	ommunity service work condition requiring the performance of munity service work for a specified number of hours (not exceeding hours),	8 9 10
	(c)		chabilitation or treatment condition requiring the offender to icipate in a rehabilitation program or to receive treatment,	11 12
	(d)	an a both	bstention condition requiring abstention from alcohol or drugs or	13 14
	(e)	a no	on-association condition prohibiting association with particular ons,	15 16
	(f)		ace restriction condition prohibiting the frequenting of or visits to a sicular place or area,	17 18
	(g)	a sup	pervision condition requiring the offender to submit to supervision:	19
		(i)	by a community corrections officer, except as provided by subparagraph (ii) or (iii), or	20 21
		(ii)	if the offender was under the age of 18 years when the condition was imposed, by a juvenile justice officer until the offender has reached that age, or	22 23 24
		(iii)	if the offender was under the age of 18 years when the condition was imposed but has since reached that age, by a juvenile justice officer where the officer chooses to continue supervision.	25 26 27
(3)			nal condition of any of the following kinds must not be imposed on ity correction order:	28 29
	(a)	a ho	me detention condition,	30
	(b)	an e	lectronic monitoring condition,	31
	(c)		rfew condition imposing a curfew exceeding 12 hours in any period 4 hours.	32 33
(4)	a co offer <b>Note</b>	mmun nder is . The  p	cing court must not impose a community service work condition on ity correction order unless an assessment report states that the suitable to be the subject of such a condition.  provisions of section 17D (4) refer specifically to assessment reports ormmunity service work conditions.	34 35 36 37 38
(5)	The	senter	ncing court may limit the period during which an additional	39
	<b>Note</b> 1999	Regul	on a community correction order is in force. lations may be made under the <i>Crimes (Administration of Sentences) Act</i> escribe the offender's obligations under an additional condition of a correction order.	40 41 42 43
Furti	ner co	nditio	ns	44
(1)		ication	ncing court may at the time of sentence, or subsequently on the of a community corrections officer or juvenile justice officer or the	45 46 47
	(a)		ose further conditions on a community correction order, or	48

		(b) vary or revoke any such further conditions imposed by it on a community correction order.	1 2
	(2)	This section does not permit the sentencing court to impose any further conditions, or vary any such further conditions, if the conditions as imposed or varied:	3 4 5
		(a) would be inconsistent with:	6
		(i) any of the standard conditions of a community correction order, or	7 8
		(ii) any of the additional conditions (whether or not actually imposed on the community service order) referred to in section 89 (2), or	9 10
		(b) would not be permitted under section 89 (3).	11
	(3)	The sentencing court may limit the period during which a further condition on a community correction order is in force.	12 13
91	Pow	er of court in dealing with applications	14
	(1)	The sentencing court may refuse to consider an application by the offender under section 89 or 90 if the court is satisfied that the application is without merit.	15 16 17
	(2)	The sentencing court may, with the consent of the community corrections officer or juvenile justice officer and the offender, deal with the application with or without parties being present and in open court or in the absence of the public.	18 19 20 21
	(3)	To avoid doubt, a court may deal with an application even though the court is constituted differently from the court as constituted at the time of the sentence.	22 23
Par	t 8	Sentencing procedures for conditional release orders	24 25
Divi	sion	1 Preliminary	26
94	Appl	ication	27
		This Part applies in circumstances in which a court is considering, or has made, a conditional release order.	28 29
Divi	sion	2 Term and commencement	30
95	Term	of conditional release order	31
	(1)	The term of a conditional release order is the period specified in the order.	32
	(2)	The maximum term of a conditional release order is 2 years.	33
96	<u> </u>		34
	Com	mencement of conditional release order	0-
	Com	A conditional release order on the date on which it is made.	35
Divi	sion	A conditional release order commences on the date on which it is made.	
Divi 97	sion	A conditional release order commences on the date on which it is made.	35
	sion	A conditional release order commences on the date on which it is made.  Conditions	35 36

		(b) any additional conditions imposed by the sentencing court under section 99,	1 2
		(c) any further conditions imposed by the sentencing court under section 99A.	3 4
		<b>Note.</b> Provisions relating to a breach of obligations under a conditional release order are contained in section 108C of the <i>Crimes (Administration of Sentences) Act</i> 1999.	5 6
98	Stan	dard conditions	7
	(1)	The sentencing court must at the time of sentence impose on a conditional release order the standard conditions of a conditional release order.	8 9
	(2)	The standard conditions of a conditional release order are the following:	10
		(a) a condition that the offender must not commit any offence,	11
		(b) a condition that the offender must appear before the court if called on to do so at any time during the term of the conditional release order.	12 13
		<b>Note.</b> Regulations may be made under the <i>Crimes (Administration of Sentences) Act</i> 1999 to prescribe the offender's obligations under a standard condition of a conditional release order.	14 15 16
99	Addi	tional conditions	17
	(1)	The sentencing court may at the time of sentence, or subsequently on the application of a community corrections officer or juvenile justice officer or the offender:	18 19 20
		(a) impose on a conditional release order any of the additional conditions of a conditional release order, or	21 22
		(b) vary or revoke any such additional conditions imposed by it on a conditional release order.	23 24
	(2)	The additional conditions of a conditional release order that are available to be imposed are the following (as directed by the sentencing court):	25 26
		(a) a rehabilitation or treatment condition requiring the offender to participate in a rehabilitation program or to receive treatment,	27 28
		(b) an abstention condition requiring abstention from alcohol or drugs or both,	29 30
		(c) a non-association condition prohibiting association with particular persons,	31 32
		(d) a place restriction condition prohibiting the frequenting of or visits to a particular place or area,	33 34
		(e) a supervision condition requiring the offender to submit to supervision:	35
		(i) by a community corrections officer, except as provided by subparagraph (ii) or (iii), or	36 37
		(ii) if the offender was under the age of 18 years when the condition was imposed, by a juvenile justice officer until the offender has reached that age, or	38 39 40
		(iii) if the offender was under the age of 18 years when the condition was imposed but has since reached that age, by a juvenile justice officer where the officer chooses to continue supervision.	41 42 43
	(3)	An additional condition of any of the following kinds must not be imposed on a conditional release order:	44 45
		(a) a home detention condition,	46
		(b) an electronic monitoring condition,	47

			(c)	a curfew condition,	1
			(d)	a community service work condition.	2
		(4)		sentencing court may limit the period during which an additional ition on a conditional release order is in force.	3 4
			1999	Regulations may be made under the <i>Crimes (Administration of Sentences) Act</i> to prescribe the offender's obligations under an additional condition of a tional release order.	5 6 7
	99A	Furti	ner co	nditions	8
		(1)		sentencing court may at the time of sentence, or subsequently on the cation of a community corrections officer or juvenile justice officer or the ider:	9 10 11
			(a)	impose further conditions on a conditional release order, or	12
			(b)	vary or revoke any such further conditions imposed by it on a conditional release order.	13 14
		(2)		section does not permit the sentencing court to impose any further itions, or vary any such further conditions, if the condition as imposed or d:	15 16 17
			(a)	would be inconsistent with:	18
				(i) any of the standard conditions of a conditional release order, or	19
				(ii) any of the additional conditions (whether or not actually imposed on the conditional release order) referred to in section 99 (2), or	20 21
			(b)	would not be permitted under section 99 (3).	22
		(3)		sentencing court may limit the period during which a further condition on additional release order is in force.	23 24
	100	Pow	er of c	ourt in dealing with applications	25
		(1)		sentencing court may refuse to consider an application by the offender r section 99 or 99A if the court is satisfied that the application is without t.	26 27 28
		(2)	office	sentencing court may, with the consent of the community corrections er or juvenile justice officer and the offender, deal with the application or without parties being present and in open court or in the absence of the ic.	29 30 31 32
		(3)		void doubt, a court may deal with an application even though the court is tituted differently from the court as constituted at the time of the sentence.	33 34
[32]	Sect	ion 10	0R Pro	oceedings for breach of order	35
	Omit	sectio	n 100I	R (2). Insert instead:	36
		(2)	If the	e offender fails to appear, the court may:	37
			(a)	issue a warrant for the offender's arrest, or	38
			(b)	authorise an authorised officer to issue a warrant for the offender's arrest.	39 40
		(2A)	befor	owever, at the time the court proposes to call on an offender to appear re it, the court is satisfied that the location of the offender is unknown, the may immediately:	41 42 43
			(a)	issue a warrant for the offender's arrest, or	44

	(b) authorise an authorised officer to issue a warrant for the offender's arrest.	1			
[33]	Section 100T Right to decide not to participate in intervention program				
	Omit "section 98 (1A) or (1B)" from section 100T (4).	4			
	Insert instead "section 100R (2) or (2A)".	5			

Schedule 2			Amendment of Crimes (Sentencing Procedure) Act 1999 No 92—savings and transitional provisions		
	Sche	edule 2	Savings, transitional and other provisions	4	
	Inser	t after	Part 28:	5	
	Par	t 29	Provisions consequent on enactment of Crimes	6	
			(Sentencing Procedure) Amendment (Sentencing Options) Act 2017	7 8	
	Divi	sion	1 Preliminary	9	
	70	Defir	iitions	10	
			In this Part:	11	
			amending Act means the Crimes (Sentencing Procedure) Amendment (Sentencing Options) Act 2017.	12 13	
			commencement day means the day appointed for the commencement of the	14	
			insertion, repeal or amendment (made by the amending Act) that is relevant to the provision in which the expression occurs.	15 16	
	Divi	sion	2 Existing bonds and orders	17	
	71	Exist	ing home detention orders under section 6	18	
		(1)	This clause applies to a home detention order (the home detention order)	19	
			made under section 6 before its repeal by the amending Act and in force immediately before the commencement day. A reference in this clause to the	20 21	
			home detention order includes a reference to the sentence of imprisonment that	22	
			is being served by way of home detention, including any period during which the offender is required to be subject to a parole order.	23 24	
		(2)			
		(2)	The home detention order is taken to be an intensive correction order ( <i>the intensive correction order</i> ) made under section 7 as substituted by the amending Act.	25 26 27	
		(3)	The sentence of imprisonment served by way of the home detention order is	28	
		. ,	taken to be a sentence of imprisonment served by way of the intensive	29	
		(4)	correction order.	30	
		(4)	The intensive correction order is on the commencement day subject only to:	31	
			<ul><li>(a) the standard conditions of an intensive correction order, and</li><li>(b) a home detention condition, and</li></ul>	32	
			(c) any conditions imposed under section 103 (1) (b) or (c) of the <i>Crimes</i>	33 34	
			(Administration of Sentences) Act 1999 and in force immediately before the commencement day in respect of the home detention order, and	35 36	
			(d) any other conditions prescribed by or determined under the regulations.	37	
		(5)	Subject to this Act and the <i>Crimes (Administration of Sentences) Act 1999</i> , the intensive correction order operates for the same term as the home detention order.	38 39 40	
		(6)	A parole order made in relation to the home detention order ceases to have effect on and from the commencement day.	41 42	

conditions of the intensive correction order only.

If a home detention order includes a period in which the offender is required

to be subject to a parole order, the period during which the offender would

have been subject to the parole order is to be served subject to the standard

1

2

3

4

(7)

	(8)	Without limitation, the regulations may make provision for or with respect to conditions that apply to intensive correction orders referred to in this clause.	;
		<b>Note.</b> Clause 125 of Schedule 5 to the <i>Crimes (Administration of Sentences) Act 1999</i> provides for the manner in which the Parole Authority is to exercise its functions under section 81A of that Act (Conditions—Parole Authority) in relation to the conditions of an intensive correction order referred to in this clause.	- 8 9
72	Exis	sting intensive correction orders under section 7	11
	(1)	This clause applies to an intensive correction order ( <i>the intensive correction order</i> ) made under section 7 before its substitution by the amending Act and in force immediately before the commencement day.	12 13 14
	(2)	The intensive correction order is taken to have been made under section 7 as substituted by the amending Act.	15 16
	(3)	The intensive correction order is on the commencement day subject only to:	17
		(a) the standard conditions of an intensive correction order, and	18
		(b) any conditions imposed under section 81 (3) of the <i>Crimes</i> (Administration of Sentences) Act 1999 and in force immediately before the commencement day in respect of the intensive order, and	19 20 21
		(c) any other conditions prescribed by or determined under the regulations including conditions for or relating to a community service work requirement.	2; 2; 24
	(4)	Sections 163 and 164 of the <i>Crimes (Administration of Sentences) Act 1999</i> as in force on or after the commencement day apply to the intensive correction order in respect of action that may be taken for a breach of the order.	25 26 27
	(5)	The conditions imposed on the intensive correction order by the court under section 81 of the <i>Crimes (Administration of Sentences) Act 1999</i> as in force before the commencement day cease to apply to the order. This subclause does not limit subclause (3) or (4).	28 29 30 31
		<b>Note.</b> Clause 125 of Schedule 5 to the <i>Crimes (Administration of Sentences) Act 1999</i> provides for the manner in which the Parole Authority is to exercise its functions under section 81A of that Act (Conditions—Parole Authority) in relation to the conditions of an intensive correction order referred to in this clause.	32 33 34 35
73	Exis	sting community service orders under section 8	36
	(1)	This clause applies to a community service order ( <i>the community service order</i> ) made under section 8 before its substitution by the amending Act and in force immediately before the commencement day.	37 38 39
	(2)	The community service order is taken to be a community correction order ( <i>the community correction order</i> ) made under section 8 as substituted by the amending Act.	40 47 42
	(3)	The community correction order is on the commencement day subject only to:	43
		(a) the standard condition referred to in section 88 (2) (b) as inserted by the amending Act (that the offender must appear before the court if called on to do so at any time during the term of the community correction order), and	44 45 46 47
		(b) a community service work condition, and	48

		(c) any other conditions prescribed by or determined under the regulations.	1
	(4)	Subject to this Act and the <i>Crimes (Administration of Sentences) Act 1999</i> , the community correction order operates for the same term as the community service order.	2 3 4
	(5)	Without limitation, the regulations may make provision for or with respect to community service work required by or under community correction orders referred to in this clause.	5 6 7
74	Exis	ting good behaviour bonds under section 9	8
	(1)	This clause applies to a good behaviour bond entered into under section 9 of this Act ( <i>the section 9 bond</i> ) before its substitution by the amending Act and in force immediately before the commencement day.	9 10 11
	(2)	The section 9 bond is taken to be a community correction order ( <i>the community correction order</i> ) made under section 8 as substituted by the amending Act.	12 13 14
	(3)	The community correction order is on conversion subject only to:	15
		(a) the standard conditions of a community correction order, and	16
		(b) any conditions imposed on the section 9 bond under section 95 (c) before the commencement day and in force immediately before that day, and	17 18 19
		(c) any other conditions prescribed by or determined under the regulations.	20
	(4)	The community correction order expires on the date set by the sentencing court before the commencement day in relation to the section 9 bond.	21 22
	(5)	If a condition relating to supervision was imposed on the section 9 bond by a court before the commencement day and was in force immediately before that day, the offender is subject to such obligations in respect of the condition as may be prescribed by regulations made under the <i>Crimes (Administration of Sentences) Act 1999</i> .	23 24 25 26 27
	(6)	Any warrant issued under section 98 in respect of the section 9 bond before the commencement day continues to have effect. The warrant authorises a police officer to arrest and bring the offender before a court to be dealt with for an alleged breach of the section 9 bond.	28 29 30 31
	(7)	An offender who is brought before a court on a warrant referred to in subclause (6) is to be dealt with under sections 107C and 107D of the <i>Crimes (Administration of Sentences) Act 1999</i> .	32 33 34
75	Exis	ting good behaviour bonds under section 10 (1) (b)	35
	(1)	This clause applies to a good behaviour bond entered into under section 10 (1) (b) ( <i>the section 10 bond</i> ) before its substitution by the amending Act and in force immediately before the commencement day.	36 37 38
	(2)	The section 10 bond is taken to be a conditional release order ( <i>the conditional release order</i> ) made under section 9 as substituted by the amending Act, without proceeding to conviction, pursuant to section 10 (1) (b) as in force on and after the commencement day.	39 40 41 42
	(3)	The conditional release order is on the commencement day subject only to:	43
		(a) the standard conditions of a conditional release order, and	44

		(b)	any condition imposed on the section 10 bond under section 95 (c) before the commencement day and in force immediately before that day, and	1 2 3
		(c)	any other conditions prescribed by or determined under the regulations.	4
	(4)		conditional release order expires on the date set by the sentencing court ore the commencement day in relation to the section 10 bond.	5 6
	(5)	before the country be p	supervision condition was imposed on the section 10 bond by a court of the commencement day and was in force immediately before that day, offender is subject to such obligations in respect of the condition as may prescribed by regulations made under the <i>Crimes (Administration of tences) Act 1999</i> .	7 8 9 10 11
	(6)	com	warrant issued under section 98 in respect of a section 10 bond before the imencement day continues to have effect. The warrant authorises a police cer to arrest and bring the offender before a court to be dealt with for an ged breach of the section 10 bond.	12 13 14 15
	(7)	subc	offender who is brought before a court on a warrant referred to in clause (6) is to be dealt with under sections 108C and 108D of the <i>Crimes ministration of Sentences</i> ) Act 1999 (variations, breach and revocations).	16 17 18
76	Exis	ting s	uspended sentence orders under section 12	19
	(1)		s clause applies to an order ( <i>the suspended sentence order</i> ) made under ion 12 before its repeal by the amending Act:	20 21
		(a)	suspending execution of a sentence of imprisonment, and	22
		(b)	directing that the offender enter into a good behaviour bond,	23
		and i	in force immediately before the commencement day.	24
	(2)	conti	repeal of section 12 and any associated provisions does not affect the inuity of operation of the suspended sentence order. The order continues orce despite the repeal of that section, subject to this Act.	25 26 27
	(3)		Act and the regulations continue to apply to and in respect of the owing as if the amending Act had not been enacted:	28 29
		(a)	the suspended sentence order and good behaviour bond,	30
		(b)	the person subject to the order and bond,	31
		(c)	action that may be taken for failure to comply with any of the conditions of the bond.	32 33
	(4)	If a c	court (under section 98 as previously in force but continuing to apply under clause (3)) revokes the good behaviour bond:	34 35
		(a)	the suspended sentence order ceases to have effect in relation to the sentence of imprisonment suspended by the order, and	36 37
		(b)	the court must either:	38
			(i) order that the offender be sentenced or re-sentenced to imprisonment to be served in full-time detention, or	39 40
			(ii) make an intensive correction order under this Act, as amended by the amending Act, in respect of the offence concerned, and	41 42
		(c)	this Act (including Part 4), as amended by the amending Act, applies to the sentencing or re-sentencing of the offender under this clause in the same way as it applies to the sentencing of an offender on a conviction, and	43 44 45 46

(d) section 24 applies to the sentencing or re-sentencing of the offender under this clause in the same way as it applies to the sentencing of an offender on a conviction, but taking into account: the fact that the offender has been the subject of the good behaviour bond, and (ii) anything done by the offender in compliance with the offender's obligations under the good behaviour bond, and the offender who under this clause is sentenced or re-sentenced by a court for an offence has the same rights of appeal as the offender would 9 have had if the offender had been sentenced by that court on being 10 convicted of the offence. 11 However, subclauses (2), (3) and (4) cease to have effect in relation to the 12 suspended sentence order at the end of the period of 3 years commencing with 13 the commencement day (unless the order had already expired or otherwise 14 ceased to be in force). 15 If the suspended sentence order was still in force immediately before the end 16 of that period of 3 years, the offender is required to appear before the court and 17 is to be re-sentenced. If the offender fails to appear, the court may issue a 18 warrant for the offender's arrest. 19 Without limitation, the regulations may make provision for or with respect to 20 the revocation of the suspended sentence order and for the re-sentencing of the 21 offender. 22 References in other legislation 23 Except in so far as the context or subject matter otherwise indicates or requires 24 or the regulations otherwise provide, a reference (however expressed) in 25 another Act, or statutory rule under another Act, to: 26 a home detention order, is taken to include a reference to an intensive 27 correction order that is subject to a home detention condition, and 28 a community service order, is taken to include a reference to a 29 community correction order, and 30 a good behaviour bond, is taken to include a reference to a community (c) 31 correction order or a conditional release order or both. 32 Variations of conditions of converted orders 33 Subject to subclause (2), nothing in clauses 71–75 prevents the imposition, 34 variation or revocation (in accordance with this Act) of conditions referred to 35 in those clauses after the commencement day. 36 However, in the case of an application made to a court (under section 89, 90, 37 99 or 99A) for the imposition, variation or revocation of a condition on: 38 the community correction order referred to in clause 73 (2) or 74 (2), or (a) 39 (b) the conditional release order referred to in clause 75 (2), 40

the court must, as far as practicable, not make an order that would result in the

conditions of the community correction order or conditional release order

being more onerous than the conditions that applied to the order immediately

before the commencement day.

1

2

3

4

5

6

7

8

41

42

43

44

(5)

(7)

77

Divi	Division 3		Periodic detention orders	1		
79	Application		n of this Part	2		
	• •	This Part applies to a periodic detention order ( <i>the periodic detention order</i> ):				
		(a)	made before the repeal of section 6 (Periodic detention) by the <i>Crimes</i> (Sentencing Legislation) Amendment (Intensive Correction Orders) Act 2010, and	3 4 5 6		
		(b)	continued in force by that Act after that repeal.	7		
80	Exis	ting p	eriodic detention orders	8		
	(1)		clause applies to the periodic detention order to which this Part applies that was in force immediately before the commencement day.	9 10		
	(2)		periodic detention order ceases to apply to the offender named in the order nd from the commencement day.	11 12		
	(3)	preso comi info	Commissioner must notify the court that sentenced the offender, or a court cribed by the regulations ( <i>the court</i> ), of that fact within 14 days mencing with the commencement day, and provide the court with such mation about the offender and the periodic detention order as may be cribed by the regulations.	13 14 15 16 17		
	(4)		being notified under subclause (3), the court may call on the offender to ar before it on a date to be fixed by the court to be re-sentenced.	18 19		
	(5)	If the	e offender fails to appear, the court may issue a warrant for the offender's st.	20 21		
(6)		When re-sentencing the offender, the court must, without limitation, take into account:				
		(a)	the gravity of the offence for which the periodic detention order was made, and	24 25		
		(b)	the offender's record of performance during periodic detention and compliance with applicable requirements of periodic detention, and	26 27		
		(c)	the offender's current circumstances.	28		
81	Exis	Existing warrants for revoked periodic detention orders				
	(1)	This	clause applies where:	30		
		(a)	a warrant (the warrant) had been issued under section 181 of the Crimes (Administration of Sentences) Act 1999 or section 26 of the Periodic Detention of Prisoners Act 1981 in respect of a periodic detention order to which this Part applies, and	31 32 33 34		
		(b)	the warrant was in force immediately before the commencement day but the periodic detention order had been revoked before that day.	35 36		
	(2)	The	warrant continues in force after the commencement day.	37		
82	Prov	Provisions regarding warrants referred to in clauses 80 and 81				
	(1)	This	clause applies to:	39		
		(a)	warrants issued under clause 80, and	40		
		(b)	warrants referred to in clause 81.	41		
	(2)		blice officer who arrests or has custody of the offender named in a warrant hich this clause applies may convey the offender to the correctional centre	42 43		

			ed in the warrant and deliver the offender into the custody of the governor at correctional centre.	1 2			
	(3)	or her custody under subclause (2) must notify the court that sentenced the offender, or a court of equivalent jurisdiction, within 14 days that the offender has been received into his or her custody.					
	(4)						
	(5)	Without limitation, the regulations may make provision for or with respect to:		10			
		(a)	the return to custody of offenders subject to warrants to which this clause applies, and	11 12			
		(b)	the re-sentencing of offenders under this clause, and	13			
		(c)	the issue and execution of warrants for those purposes.	14			
Division 4 Other provisions							
83	Court to take community service order or good behaviour bond into account when sentencing						
		In se	entencing an offender, the court must, under section 24, also take into unt:	18 19			
		(a)	in the case of an offender who is being sentenced as a result of failing to comply with the offender's obligations under a community service order or good behaviour bond to which a clause of Division 2 applies:	20 21 22			
			(i) the fact that the person has been the subject of such an order or bond, and	23 24			
			(ii) anything done by the offender in compliance with the offender's obligations under the order or bond, and	25 26			
		(b)	in the case of an offender who is being sentenced as a result of deciding not to participate in, or to continue to participate in a good behaviour bond to which a clause of Division 2 applies, anything done by the offender in compliance with the offender's obligations under the good behaviour bond.	27 28 29 30 31			
84	Correction and adjustment of sentences						
	(1)	A ref	ference in section 43 to imposing a sentence extends to the making of a e detention order, a community service order, an order under section 12 or	33 34			
	(2)	Secti and	on 43 extends to authorising the court to sentence the offender, or convict sentence the offender, in accordance with this Act as amended by the	35 36 37			
		amei	nding Act, as if the order had not been made.	38			
85	Procedure following failure to enter into good behaviour bond						
	•		person:	40			
		(a)	was the subject of an order under section 9 (as in force before the date of the substitution of that section by the amending Act) directing the person to enter into a good behaviour bond, and:	41 42 43			
			(i) the order was in force immediately before that date, and	44			
			(ii) the person had not entered into such a bond in accordance with the order before that date, or	45 46			

(b) was the subject of an order under section 10 (1) (b) (in force immediately before the date of the substitution of that paragraph by the amending Act) discharging the person on condition that the person enter into a good behaviour bond, and: the order was in force immediately before that date, and the person had not entered into such a bond in accordance with (ii) the order before that date, the court may sentence the offender, or convict and sentence the offender, in accordance with this Act as amended by the amending Act, as if the order had not been made. Appeals in respect of converted order and bonds This clause applies to an order or bond (an original order or bond) referred to in clause 71 (1), 72 (1), 73 (1), 74 (1) or 75 (1). Note. Clauses 71–75 deal respectively with existing home detention orders, intensive correction orders, community service orders, section 9 bonds and section 10 bonds, which were made or entered into before the commencement day. An appeal in respect of an original order or bond that was pending immediately before the commencement day is not affected by the amending Act. However: if the court hearing the appeal sentences or re-sentences the offender, it must do so in accordance with this Act as in force on and after the commencement day, or to the extent the court confirms the original order or bond, this clause does not affect the operation of whichever of clauses 71–75 is relevant to the order. An appeal may be made in respect of an original order or bond on or after the commencement day as if the amending Act had not been enacted. However:

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

39

40

41

42

43

44

45

46

47

48

49

- (3)
  - if the court hearing the appeal sentences or re-sentences the offender, it must do so in accordance with this Act as in force on and after the commencement day, or
  - to the extent the court confirms the original order or bond, this clause does not affect the operation of whichever of clauses 71–75 is relevant to the order.

Note. This subclause does not confer a right of appeal where a right of appeal did not exist before the commencement day.

## 87 Regulations

86

- The regulations made under clause 1 in relation to the amending Act or under another clause of this Part have effect despite anything to the contrary in this Part.
- To avoid doubt, the regulations made under clause 1 may be made in relation (2) to amendments made by Schedules 1–4 to the amending Act. Nothing in this subclause affects any power under another Act that is amended by Schedule 3 or 4 to the amending Act to make regulations of a savings or transitional nature.
- The regulations made under clause 1 may make separate savings and transitional provisions or amend this Part to consolidate the savings and transitional provisions.
- (4) This clause does not affect the meaning or construction of any other Part of this Schedule.

Scl	nedu	le 3		Amendment of Crimes (Administration of Sentences) Act 1999 No 93	1
[1]	Sect	ion 3 I	nterpr	retation	3
				ions of community service order, home detention order and ICO mmittee from section 3 (1).	5
[2]	Sect	ion 3 (	(1)		6
	Inser	t in alp	habeti	ical order:	7
			Crim	munity correction order means an order in force under section 8 of the nes (Sentencing Procedure) Act 1999 or section 79 of the Fines Act 1996.	9
				ditional release order means an order in force under section 9 of the des (Sentencing Procedure) Act 1999.	10 11
[3]	Sect	ion 40	Certa	in unlawful absences not to affect length of sentence	12
	Omit	", hor	ne det	ention order" from section 40 (1) (c).	13
[4]	Part	3 Impi	risonn	nent by way of intensive correction in the community	14
	Omit	the in	troduc	etory note.	15
[5]	Sect	ions 8	1-82A	L Company of the Comp	16
	Omit	section	ns 81	and 82. Insert instead:	17
	81	Cond	ditions	s governing intensive correction orders	18
			Conc	ditions on an intensive correction order:	19
			(a)	are imposed at the time of sentence by the sentencing court under the <i>Crimes (Sentencing Procedure) Act 1999</i> (see Part 5 of that Act), and	20 21
			(b)	may be imposed, varied or revoked by the Parole Authority under this Act (see sections 81A and 164 of this Act).	22 23
	81A	Cond	ditions	s—Parole Authority	24
		(1)		Parole Authority may, on the application of a community corrections er or the offender:	25 26
			(a)	impose any conditions on an intensive correction order, or	27
			(b)	vary or revoke any conditions of an intensive correction order, including conditions imposed by the sentencing court.	28 29
		(2)	The	Parole Authority must not:	30
			(a)	vary or revoke a standard condition under this section, or	31
			(b)	impose a condition under this section unless the sentencing court could have imposed the condition under Part 5 of the <i>Crimes (Sentencing Procedure) Act 1999</i> , or	32 33 34
			(c)	vary a condition under this section unless the sentencing court could have imposed the varied condition under that Part, or	35 36
			(d)	impose a condition imposing a period of home detention or a condition requiring community service work unless a report prepared by a community corrections officer states that the imposition of such a condition is appropriate in the circumstances.	37 38 39 40

	(3)	If the Parole Authority revokes an additional concorrection order, it must replace it with another add there is already another additional condition in force	itional condition, unless	1 2 3
82	Adm	inistration of intensive correction orders and oblig	ations of offenders	4
	(1)	The regulations may make provision for or with resp	pect to:	5
		(a) the administration of intensive correction conditions of those orders, whether by the establishment or otherwise, and		6 7 8
		(b) without limitation:		9
		<ul> <li>(i) any matter relating to carrying out or co conditions of an intensive correction example, conditions relating to super electronic monitoring, curfews and con and</li> </ul>	order (including, for vision, home detention, mmunity service work),	10 11 12 13 14
		(ii) the functions of persons involved in intensive correction orders, and		15 16
		(iii) the manner in which an offender's fail offender's obligations under an intensive dealt with, and	correction order may be	17 18 19
		(iv) the service of notices on an offender.	:	20
	(2)	The obligations of an offender under an intensive co the obligations of an offender under a condition of prescribed by the regulations.	the order, are to be as	21 22 23
82A	Susp	ension of certain conditions of intensive correctio	n order	24
	(1)	This section applies to the following conditions of order:		25 26
		(a) a condition referred to in section 73 (2) (b) of <i>Procedure</i> ) Act 1999 (a supervision condition	the Crimes (Sentencing),	27 28
		(b) a curfew condition, non-association condition referred to in section 73A (2) of that		29 30
	(2)	A community corrections officer may, by order in w regulations, suspend the application of a supervision for a period or periods or indefinitely.	condition to an offender	31 32 33
	(3)	A community corrections officer may, by order giver subject to the regulations, suspend the application of which this section applies to an offender for a period	f any other condition to	34 35 36
	(4)	The suspension may be unconditional or subject to co	onditions.	37
	(5)	For the purposes of this Act, a failure to comply suspension is taken to be a failure to comply with intensive correction order. This subsection does no community corrections officer to revoke the suspension	t the obligations of the t limit the power of the	38 39 40 41
	(6)	The regulations may make provision for or with resp	ect to the following:	42
		(a) the periods for which a condition may be suspe		43
		(b) notice to an offender of the making or revocation	•	44
		(c) requiring the power to make, amend or revoke exercised subject to and in accordance with any		45 46

[6]	Sect	ion 83				1
	Omit	the se	ction.	Insert i	nstead:	2
	83	Dura	tion o	f inten	sive correction order	3
			Unle	ss soon of the te	er revoked, an offender's intensive correction order expires at the erm of the sentence or sentences to which it relates.	4
[7]		3, Divi iremer		2 Permi	ission for non-compliance with work or reporting	6
	Omit	the Di	vision	1.		8
[8]	Part	3, Divi	sion 3	Breac	ch of intensive correction order	9
	Inser	t after	Note	. Sectior ctions o	ns 163 and 164 deal with the actions that may be taken by a community fficer and the Parole Authority for breaches of an intensive correction	10 11 12 13
[9]	Sect	ions 89	9 and	90		14
	Omit	the se	ctions			15
[10]	Sect	ion 92	ICO N	/lanage	ment Committee	16
	Omit	the se	ction.			17
[11]	Sect	ion 93	Regu	lations		18
	Omit	the se	ction.			19
[12]	Part	4 Impr	isonn	nent by	way of home detention	20
	Omit	the Pa	rt.			21
[13]	Parts	s 4B ar	nd 4C			22
	Inser	t after !	Part 4.	A:		23
	Par	t 4B	Adı	minis	stration of community correction orders	24
	107A	Conc	litions	gover	ning community correction orders	25
			cour	t under	on a community correction order are imposed by the sentencing the <i>Crimes (Sentencing Procedure) Act 1999</i> (see Part 7 of that court under section 107D.	26 27 28
	107B	Admi	inistra	ation of	community correction orders and obligations of offenders	29
		(1)	The	•	ons may make provision for or with respect to:	30
			(a)	condi	dministration of community correction orders, including any tions of those orders, whether by the establishment of a scheme for management or otherwise, and	31 32 33
			(b)		ut limitation:	34
				(i)	any matter relating to carrying out or complying with any of the conditions of a community correction order (including, for example, conditions relating to supervision, curfews and community service work), and	35 36 37 38
				(ii)	the functions of persons involved in the administration of community correction orders, and	39 40

		(iii) the manner in which an offender's failure to comply with the offender's obligations under a community correction order may be dealt with, and	2
		(iv) the service of notices on an offender.	4
	(2)	The obligations of an offender under a community correction order, including the obligations of an offender under a condition of the order, are to be as prescribed by the regulations.	( -
107C	Brea	ach of community correction order	8
	(1)	If it suspects that an offender may have failed to comply with any of the conditions of a community correction order:	10
		(a) the court that made the order, or	11
		(b) any other court of like jurisdiction, or	12
		(c) with the offender's consent, any other court of superior jurisdiction,	13
		may call on the offender to appear before it.	14
	(2)	If the offender fails to appear, the court may:	15
		(a) issue a warrant for the offender's arrest, or	16
		(b) authorise an authorised officer to issue a warrant for the offender's arrest.	17 18
	(3)	If, however, at the time the court proposes to call on an offender to appear before it, the court is satisfied that the location of the offender is unknown, the court may immediately:	19 20 21
		(a) issue a warrant for the offender's arrest, or	22
		(b) authorise an authorised officer to issue a warrant for the offender's arrest.	23 24
	(4)	For the purposes of subsection (1) (c), a court is of superior jurisdiction to the court that made the community correction order if it is a court to which the offender has (or has had) a right of appeal in respect of the conviction or sentence from which the order arises.	25 26 27 28
	(5)	If it is satisfied that an offender appearing before it has failed to comply with any of the conditions of a community correction order, a court:	29 30
		(a) may decide to take no action in respect of the failure to comply, or	31
		(b) may vary or revoke any conditions of the order (other than standard conditions) or impose further conditions on the order, or	32 33
		(c) may revoke the order.	34
	(6)	To avoid doubt, the court that made an order may deal with a breach of a community correction order even though the court is constituted differently from the court as constituted at the time the order was made.	35 36 37
	(7)	In this section:	38
		authorised officer has the same meaning as it has in the Criminal Procedure Act 1986.	39 40
107D	Con	sequences of revocation of community correction order	4
	(1)	If a court revokes a community correction order, it may re-sentence the offender for the offence to which the order relates.	42 43

	(2)	The <i>Crimes (Sentencing Procedure) Act 1999</i> applies to the re-sentencing of an offender under this section in the same way as it applies to the sentencing of an offender on a conviction.	2
	(3)	An offender who under this section is re-sentenced by a court for an offence has the same rights of appeal as the offender would have had if the offender had been sentenced by that court on being convicted of the offence.	!
107E	Susp	pension of certain conditions of community correction order	7
	(1)	This section applies to the following conditions of a community correction order:	9
		(a) a condition referred to in section 89 (2) (g) of the <i>Crimes (Sentencing Procedure) Act 1999</i> (a <i>supervision condition</i> ),	10 17
		(b) a curfew condition, non-association condition or place restriction condition referred to in section 89 (2) of that Act.	12 13
	(2)	A community corrections officer may, by order in writing and subject to the regulations, suspend the application of a supervision condition to an offender for a period or periods or indefinitely.	14 15 16
	(3)	A community corrections officer may, by order given orally or in writing and subject to the regulations, suspend the application of any other condition to which this section applies to an offender for a period or periods.	17 18 19
	(4)	The suspension may be unconditional or subject to conditions.	20
	(5)	For the purposes of this Act, a failure to comply with a condition of the suspension is taken to be a failure to comply with the obligations of the community correction order. This subsection does not limit the power of the community corrections officer to revoke the suspension order.	21 22 23 24
	(6)	The regulations may make provision for or with respect to the following:	25
		(a) the periods for which a condition may be suspended,	26
		(b) notice to an offender of the making or revocation of a suspension order,	27
		(c) requiring the power to make, amend or revoke a suspension order to be exercised subject to and in accordance with any specified requirements.	28 29
	(7)	A reference in this section to a community corrections officer is, in relation to an offender who is subject to supervision or management by a juvenile justice officer, taken to be a reference to a juvenile justice officer.	30 37 32
	(8)	If a supervision condition is suspended by a juvenile justice officer and the offender has reached the age of 18 years, a community corrections officer may revoke the suspension order.	33 34 38
Pai	t 4C	Administration of conditional release orders	36
108A	Cond	ditions governing conditional release orders	37
		Conditions on a conditional release order are imposed by the sentencing court under the <i>Crimes (Sentencing Procedure) Act 1999</i> (see Part 8 of that Act) or by a court under section 108D.	38 39 40

108B	Adm	ninistration of conditional release orders and obligations of offenders	1
	(1)	The regulations may make provision for or with respect to:	2
		(a) the administration of conditional release orders, including any conditions of those orders, whether by the establishment of a scheme for their management or otherwise, and	3 4 5
		(b) without limitation:	6
		(i) any matter relating to carrying out or complying with any of the conditions of a conditional release order (including, for example, a condition relating to supervision), and	7 8 9
		(ii) the functions of persons involved in the administration of conditional release orders, and	10 11
		(iii) the manner in which an offender's failure to comply with the offender's obligations under a conditional release order may be dealt with, and	12 13 14
		(iv) the service of notices on an offender.	15
	(2)	The obligations of an offender under a conditional release order, including the obligations of an offender under a condition of the order, are to be as prescribed by the regulations.	16 17 18
108C	Brea	ach of conditional release order	19
	(1)	If it suspects that an offender may have failed to comply with any of the conditions of a conditional release order:	20 21
		(a) the court that made the order, or	22
		(b) any other court of like jurisdiction, or	23
		(c) with the offender's consent, any other court of superior jurisdiction,	24
		may call on the offender to appear before it.	25
	(2)	If the offender fails to appear, the court may:	26
		(a) issue a warrant for the offender's arrest, or	27
		(b) authorise an authorised officer to issue a warrant for the offender's arrest.	28 29
	(3)	If, however, at the time the court proposes to call on an offender to appear before it, the court is satisfied that the location of the offender is unknown, the court may immediately:	30 31 32
		(a) issue a warrant for the offender's arrest, or	33
		(b) authorise an authorised officer to issue a warrant for the offender's arrest.	34 35
	(4)	For the purposes of subsection (1) (c), a court is of superior jurisdiction to the court that made the conditional release order if it is a court to which the offender has (or has had) a right of appeal in respect of the conviction or sentence from which the order arises.	36 37 38 39
	(5)	If it is satisfied that an offender appearing before it has failed to comply with any of the conditions of a conditional release order, a court:	40 41
		(a) may decide to take no action in respect of the failure to comply, or	42
		(b) may vary or revoke any conditions of the order (other than standard conditions) or impose further conditions on the order, or	43 44
		(c) may revoke the order.	45

	(6)	To avoid doubt, the court that made an order may deal with a breach of a conditional release order even though the court is constituted differently from the court as constituted at the time the order was made.	1 2 3
	(7)	In this section:	4
		authorised officer has the same meaning as it has in the Criminal Procedure Act 1986.	5 6
108D	Con	sequences of revocation of conditional release order	7
	(1)	If a court revokes a conditional release order, it may sentence or re-sentence the offender for the offence to which the order relates.	8 9
	(2)	The Crimes (Sentencing Procedure) Act 1999 applies to the sentencing or re-sentencing of an offender under this section in the same way as it applies to the sentencing of an offender found guilty of the offence concerned.	10 11 12
	(3)	An offender who under this section is sentenced by a court for an offence has the same rights of appeal as the offender would have had if the offender had been sentenced by that court on being found guilty of the offence concerned.	13 14 15
108E	Sus	pension of certain conditions of conditional release order	16
	(1)	This section applies to the following conditions of a conditional release order:	17
		(a) a condition referred to in section 99 (2) (e) of the <i>Crimes (Sentencing Procedure) Act 1999</i> (a <i>supervision condition</i> ),	18 19
		(b) a non-association condition or place restriction condition referred to in section 99 (2) of that Act.	20 21
	(2)	A community corrections officer may, by order in writing and subject to the regulations, suspend the application of a supervision condition to an offender for a period or periods or indefinitely.	22 23 24
	(3)	A community corrections officer may, by order given orally or in writing and subject to the regulations, suspend the application of any other condition to which this section applies to an offender for a period or periods.	25 26 27
	(4)	The suspension may be unconditional or subject to conditions.	28
	(5)	For the purposes of this Act, a failure to comply with a condition of the suspension is taken to be a failure to comply with the obligations of the conditional release order. This subsection does not limit the power of the community corrections officer to revoke the suspension order.	29 30 31 32
	(6)	The regulations may make provision for or with respect to the following:	33
		(a) the periods for which a condition may be suspended,	34
		(b) notice to an offender of the making or revocation of a suspension order,	35
		(c) requiring the power to make, amend or revoke a suspension order to be exercised subject to and in accordance with any specified requirements.	36 37
	(7)	A reference in this section to a community corrections officer is, in relation to an offender who is subject to supervision or management by a juvenile justice officer, taken to be a reference to a juvenile justice officer.	38 39 40
	(8)	If a supervision condition is suspended by a juvenile justice officer and the offender has reached the age of 18 years, a community corrections officer may revoke the suspension order.	41 42 43
4] Par	t 5 Cor	mmunity service work and other work performed by offenders	44
Om	it the ir	ntroductory note before Division 1.	45

[15]	Part	5			1
	Omi	t Divis	ion 1 a	and the heading to Division 2.	2
[16]	Sect	ion 11	8 Defi	initions	3
	Omi	t parag	raphs	(c) and (d) of the definition of <i>community service work</i> . Insert instead:	4
			(d)	community service work performed by an offender under a community correction order, and	5 6
[17]	Sect	ion 11	8, defi	inition of "offender"	7
	Omi	t parag	raphs	(c) and (d). Insert instead:	8
			(d)	a community correction order, or	9
[18]	Sect	ions 1	19 and	d 123	10
	Omi	t "a coi	nmun	ity service order" wherever occurring.	11
	Inser	t instea	ad "an	intensive correction order or community correction order".	12
[19]	Sect	ions 1	63–16	64 <b>A</b>	13
	Omi	t sectio	ns 163	3 and 164. Insert instead:	14
	163	Action orde		y community corrections officer on breach of intensive correction	15 16
		(1)	is sa	section applies if the Commissioner or a community corrections officer atisfied that an offender has failed to comply with the offender's gations under an intensive correction order.	17 18 19
		(2)	A co	ommunity corrections officer may take any of the following actions:	20
			(a)	record the breach and take no further action,	21
			(b)	give an informal warning to the offender,	22
			(c)	give, or arrange to be given to, the offender a formal warning that further breaches will result in referral to the Parole Authority,	23 24
			(d)	give a reasonable direction to the offender relating to the kind of behaviour by the offender that caused the breach,	25 26
			(e)	impose a curfew on the offender of up to 12 hours in any 24-hour period.	27
		(3)	or a c Auth recor	on alternative, or in addition, to taking any such action, the Commissioner community corrections officer may decide to refer the breach to the Parole nority because of the serious nature of the breach and may also make a mmendation as to the action that the Parole Authority may take in respect the offender.	28 29 30 31 32
		(4)	offer offic	deciding whether and what action should be taken in respect of the inder's breach of the intensive correction order, a community corrections have regard to any action previously taken in respect of the breach my earlier breaches of the order.	33 34 35 36
		(5)		regulations may make provision for or with respect to any action that may aken by a community corrections officer under this section.	37 38
	164	Actio	ons by	y Parole Authority on breach of intensive correction order	39
		(1)		section applies if the Parole Authority is satisfied that an offender has d to comply with the offender's obligations under an intensive correction r.	40 41 42

(2)	The	Parole Authority may take any of the following actions:	1
	(a)	record the breach and take no further action,	2
	(b)	give a formal warning to the offender,	3
	(c)	impose any conditions on the intensive correction order,	4
	(d)	vary or revoke any conditions of the intensive correction order, including conditions imposed by the sentencing court,	5 6
	(e)	make an order revoking the intensive correction order (a <i>revocation order</i> ).	7 8
(3)		nout affecting the generality of subsection (2), the Parole Authority may ose a condition on the intensive correction order of the following kind:	9 10
	(a)	a condition imposing a requirement that the offender remain at the offender's place of residence for a period of home detention of up to 30 days,	11 12 13
	(b)	a condition that the offender must submit to the use of an electronic monitoring device,	14 15
	(c)	ancillary conditions relating to any condition imposed under this section.	16 17
(4)	offer have	eciding whether and what action should be taken in respect of the nder's breach of the intensive correction order, the Parole Authority may regard to any action previously taken in respect of the breach or any er breaches of the order.	18 19 20 21
(5)	(Sen unde	regulations may provide that specified provisions of Part 5 of the <i>Crimes</i> tencing <i>Procedure</i> ) Act 1999 and Part 3 of this Act (and the regulations or those Acts) so far as they relate to conditions of an intensive correction or that is subject to a condition of home detention imposed under this on:	22 23 24 25 26
	(a)	apply to the offender in the same way as they apply to an intensive correction order that is subject to a home detention condition, and	27 28
	(b)	so apply with any necessary modifications and any modifications prescribed by the regulations.	29 30
(6)	The	Parole Authority must not:	31
	(a)	vary or revoke a standard condition under this section, or	32
	(b)	impose a condition under this section unless the sentencing court could have imposed the condition under Part 5 of the <i>Crimes (Sentencing Procedure) Act 1999</i> , or	33 34 35
	(c)	vary a condition under this section unless the sentencing court could have imposed the varied condition under that Part, or	36 37
	(d)	impose a condition imposing a period of home detention or a condition requiring community service work unless a report prepared by a community corrections officer states that the imposition of such a condition is appropriate in the circumstances.	38 39 40 41
(7)	corre	re Parole Authority revokes an additional condition on an intensive ection order, it must replace it with another additional condition, unless is already another additional condition in force with respect to the order.	42 43 44

	164A	Effec	ct of re	evocation order	1
		(1)	takeı	vocation order (see section 164 (2) (e)) takes effect, or is taken to have n effect, on the date on which it is made or on such earlier date as the le Authority thinks fit.	2 3 4
		(2)	the f	earliest date on which the revocation order may take effect is the date of irst occasion on which it appears to the Parole Authority that the offender d to comply with the offender's obligations under the intensive correction r.	5 6 7 8
		(3)	revo	n offender is not taken into custody until after the day on which the cation order takes effect, the term of the offender's sentence is, by this ection, extended by the number of days the person was at large after the r took effect.	9 10 11 12
[20]	Secti	ion 16	5 Parc	ole Authority may reinstate revoked intensive correction order	13
	Omit	sectio	n 165	(1) (a). Insert instead:	14
			(a)	on its own initiative or on the application of the offender, and	15
[21]	Secti	ion 16	5 (6)		16
	Inser	t after	section	n 165 (5):	17
		(6)	The	regulations may make provision for or with respect to:	18
			(a)	requiring a report to be made in relation to an assessment referred to in subsection (3), and	19 20
			(b)	the matters to be addressed in the report, and	21
			(c)	the preparation and furnishing of the report.	22
[22]	Secti	ions 1	<b>65A-</b> 1	165C	23
	Omit	the se	ctions		24
[23]	Part	7, Divi	ision 2	2 Home detention orders	25
	Omit	the D	ivisior	1.	26
[24]	Secti	ion 17	3 Noti	ice of revocation	27
	Omit	", hor	ne det	ention order" wherever occurring.	28
[25]	Secti	ion 17	4 Rev	iew of revocation	29
• •	Omit	", hor	ne det	ention order" wherever occurring.	30
[26]	Secti	ion 17	5 Dec	ision after review	31
[]				ention order" wherever occurring.	32
[27]		on 17		·	33
[-,]		"or".	• (1A)		34
[28]	Secti	ion 17	5 (1A)	ı (c)	35
[-0]		the pa	, ,		36
[29]		_		eview not available in certain circumstances	37
[20]				ention order".	38

[30]	Section 176	S Application to Supreme Court by offender	1				
	Omit ", hom	ne detention order" wherever occurring in section 176 (1).	2				
[31]	Section 179	Consequential revocation of other orders	3				
	Omit ", hom	ne detention order" and ", home detention orders" wherever occurring.	4				
[32]	Authority to	A Revocation of first of consecutive home detention orders—Parole seek new assessment	5				
	Omit the sec	ction.	7				
[33]		Warrants committing offenders to correctional centres	8				
	Omit section	n 181 (1). Insert instead:	9				
	(1)	If the Parole Authority revokes an intensive correction order or parole order, it may issue a warrant committing the offender to a correctional centre to serve the remainder of the sentence to which the order relates by way of full-time detention.	10 11 12 13				
[34]	Section 181	l (1A)	14				
	Omit the sul	osection.	15				
[35]	Section 181 (3) (b)						
		bending the Parole Authority's decision as to whether or not to make a home der under section 165A, as the case requires".	17 18				
[36]	Section 182	2 Functions may be exercised after order has expired	19				
	Omit ", hom	ne detention order".	20				
[37]	Section 185	5 Functions of Parole Authority	21				
	Omit ", hom	ne detention orders" from section 185 (1) (b).	22				
[38]	Section 193	BC Parole Authority decisions	23				
	Omit ", hom	ne detention order" from section 193C (1) (b).	24				
[39]	Section 193	3C (1) (c)	25				
	Omit "or ho	me detention order".	26				
[40]	Section 193	3C (1) (c)	27				
	Omit "or 16	7 (1)".	28				
[41]	Section 235	5E Functions of community corrections field officers	29				
	Omit "comm	nunity service orders" from section 235E (1).	30				
	Insert instea	d "community correction orders".	31				
[42]	Section 236	6M Accommodation of offenders in residential facilities	32				
	Omit ", a ho	ome detention order" from section 236M (5) (b).	33				
[43]	Section 254	Extension of sentence following unlawful absence from custody	34				
	Omit ", hom	ne detention order" from section 254 (2) (a).	35				

[44]	Section 255A Approvals for the purposes of the Mutual Assistance in Criminal Matters Act 1987 of the Commonwealth						
			is the subject of a home detention order" from paragraph (a) of the definition of <i>ficer</i> in section 255A (4).	3 4			
[45]	Sect	ion 25	9 Service of notices	5			
		t "hor on 259	me detention order, parole order or community service order" from (1).	6 7			
	Inser	t instea	ad "parole order or community correction order".	8			
[46]	Sect	ion 26	0 Evidentiary certificates	9			
	Omi	t "hom	e detention order, community service order" wherever occurring.	10			
	Inser	t instea	ad "community correction order".	11			
[47]	Sch	edule 3	3 Intensive Correction Orders Management Committee	12			
	Omi	t the So	chedule.	13			
[48]	Sche	edule (	5 Savings, transitional and other provisions	14			
	Inser	t at the	e end of the Schedule:	15			
	Par	t 22	Provisions consequent on enactment of Crimes (Sentencing Procedure) Amendment (Sentencing Options) Act 2017	16 17 18			
	124	Defir	nition and operation of this Part	19			
		(1)	In this Part:	20			
			amending Act means the Crimes (Sentencing Procedure) Amendment (Sentencing Options) Act 2017.	21 22			
			<b>commencement day</b> means the day appointed for the commencement of the insertion, repeal or amendment (made by the amending Act) that is relevant to the provision in which the expression occurs.	23 24 25			
		(2)	Nothing in this Schedule affects the operation of Part 29 of Schedule 2 to the <i>Crimes (Sentencing Procedure) Act 1999</i> in relation to amendments made to this Act by the amending Act.	26 27 28			
	125	Parole Authority's power under section 81A regarding certain conditions					
		(1)	This clause applies to an intensive correction order referred to in clause 71 (2) or 72 (2) of Schedule 2 to the <i>Crimes (Sentencing Procedure) Act 1999</i> .	30 31			
		(2)	In exercising its functions under section 81A in relation to the intensive correction order, the Parole Authority must, as far as practicable, not exercise those functions in a way that would result in the conditions of the order being more onerous than the conditions that applied to the order immediately before the commencement day.	32 33 34 35 36			
	126	Exis	ting home detention orders under section 165A	37			
		(1)	This clause applies to a home detention order ( <i>the home detention order</i> ) made by the Parole Authority under section 165A before the commencement day and in force immediately before that day.	38 39 40			

	(2)	Clause 71 of Schedule 2 to the <i>Crimes (Sentencing Procedure) Act 1999</i> (as inserted by the amending Act) applies to the home detention order made by the Parole Authority in the same way as it applies to a home detention order made by a sentencing court.	1 2 3 4
127	Tem	porary release order under section 165B	5
	(1)	This clause applies to a temporary release order (the <i>temporary release order</i> ) made under section 165B before the commencement day and in force immediately before that day.	6 7 8
	(2)	The temporary release order continues in force on and after the commencement day, pending the Parole Authority's decision as to whether or not to reinstate the intensive correction order.	9 10 11
	(3)	If the Parole Authority decides to reinstate the intensive correction order, the following conditions apply to the order:	12 13
		(a) the standard conditions of an intensive correction order in section 73 of the <i>Crimes (Sentencing Procedure) Act 1999</i> ,	14 15
		(b) any other conditions prescribed by or determined under the regulations.	16
	(4)	The Parole Authority may impose further conditions on the intensive correction order that are not inconsistent with the standard conditions and may vary or revoke any conditions of the order (other than standard conditions).	17 18 19
128	Reg	ulations	20
	(1)	The regulations made under clause 1 in relation to the amending Act or under another clause of this Part have effect despite anything to the contrary in this Part.	21 22 23
	(2)	The regulations made under clause 1 may make separate savings and transitional provisions or amend this Part to consolidate the savings and transitional provisions.	24 25 26
	(3)	This clause does not affect the meaning or construction of any other Part of this Schedule.	27 28

Sch	nedule 4 Amendment of other legislation	1					
4.1	Anzac Memorial (Building) Act 1923 No 27	2					
	Section 9 By-laws						
	Omit section 9 (4A) (a). Insert instead:						
	(a) may make a community correction order under section 8 of the <i>Crimes</i> (Sentencing Procedure) Act 1999 that is subject to the standard conditions of a community correction order and to a community service work condition (despite the offence not being punishable by imprisonment), or	5 6 7 8 9					
4.2	Bail Act 2013 No 26	10					
[1]	Section 18 Matters to be considered as part of assessment	11					
	Omit section 18 (1) (f) (v). Insert instead:	12					
	(v) home detention orders, good behaviour bonds or community service orders,	13 14					
[2]	Section 18 (1) (f) (vii) and (viii)	15					
	Omit the subparagraphs. Insert instead:	16					
	(vii) community correction orders,	17					
	(viii) conditional release orders,	18					
[3]	Schedule 1 Application of Act to non-offenders						
	Omit clause 1 (2) (a). Insert instead:						
	(a) proceedings under the <i>Crimes (Sentencing Procedure) Act 1999</i> for an alleged failure by the person to comply with the conditions of a community correction order or conditional release order imposed for an offence,	21 22 23 24					
[4]	Schedule 1, clause 1 (2) (c)						
	Omit the paragraph. Insert instead:	26					
	(c) proceedings under the <i>Crimes (Administration of Sentences) Act 1999</i> on an application for the revocation of a community correction order made in relation to the person for an offence,	27 28 29					
4.3	Births, Deaths and Marriages Registration Act 1995 No 62						
	Section 25F Definitions	31					
	Omit "or home detention order" from paragraph (b) of the definition of <i>supervision order</i> .						
4.4	Child Protection (Offenders Registration) Act 2000 No 42	33					
[1]	Section 3 Definitions	34					
	Insert in alphabetical order in section 3 (1):	35					
	community correction order means a community correction order under the Crimes (Sentencing Procedure) Act 1999.	36 37					
	conditional release order means a conditional release order under the Crimes (Sentencing Procedure) Act 1999.	38 39					

[2]	Section 3 (1), pa	ragraph (b) of the definition of "supervised sentence"	1					
	Insert "or commu	unity correction order" after "community service order".	2					
[3]	Section 3 (1), pa	ragraph (c) of the definition of "supervised sentence"	3					
	Insert "or condition	onal release order" after "good behaviour bond".	4					
[4]	Section 3 (1A)		5					
	Insert after section 3 (1):							
	or as this Act this of re Note orde	ept in so far as the context or subject-matter otherwise indicates or requires the regulations provide that this subsection not have effect, a reference in Act to the <i>Crimes (Sentencing Procedure) Act 1999</i> is a reference to that as in force at any time (whether before, on or after the commencement of subsection). This subsection does not affect the meaning or construction afterences to any other Act mentioned in this Act.  The principal purpose of this subsection is to make it clear that references to any and bonds under the <i>Crimes (Sentencing Procedure) Act 1999</i> extend to orders bonds under that Act as in force at any time.	7 8 9 10 11 12 13 14 15					
4.5	Child Protect	ion (Offenders Registration) Regulation 2015	16					
	Clause 4 Definition of "supervising authority"							
	Omit "a home detention order, community service order" from clause 4 (b) (iv).							
	Insert instead "a release order".	n intensive correction order, community correction order, conditional	19 20					
4.6	Child Protect	ion (Working with Children) Act 2012 No 51	21					
	Section 26 Person	ons not entitled to apply for review or enabling order	22					
	Insert after section	n 26 (2) (b) (i):	23					
		(ia) a community correction order or conditional release order under that Act,	24 25					
4.7	Children (Crir	ninal Proceedings) Act 1987 No 55	26					
	Section 33C App	olication of Crimes (Sentencing Procedure) Act 1999 to children	27					
	Insert ", commur bond" where first	nity correction order or conditional release order" after "good behaviour ly occurring in section 33C (1) (d).	28 29					
4.8	Civil Liability	Act 2002 No 22	30					
[1]	Section 26A Def	initions	31					
	Omit paragraph (	c) of the definition of <i>offender in custody</i> or <i>offender</i> in section 26A (1).	32					
	Insert instead:		33					
	(c)	a person in respect of whom an intensive correction order that is subject to a home detention condition is in force under the <i>Crimes (Sentencing Procedure) Act 1999</i> ,	34 35 36					

[2]	Section 26	<b>4</b> (1), paragra	aph (e) of the definition of "offender in custody" or "offender"	1			
	Omit the pa	ragraph. Inse	ert instead:	2			
		place order order <i>Order</i>	son performing community service work under, or attending a in compliance with the requirements of, an intensive correction a community correction order, or a children's community service as provided by the CAS Act or the <i>Children (Community Service rs) Act 1987</i> , whether or not the person is an offender in custody any other paragraph of this definition,	3 4 5 6 7 8			
[3]	Schedule 5	Savings, tra	ansitional and other provisions	9			
	Insert at the	end of the So	chedule, with appropriate Part and clause numbering:	10			
	Part	(Senten	ons consequent on enactment of Crimes cing Procedure) Amendment (Sentencing ) Act 2017	11 12 13			
	Defir	ition		14			
		In this Part:		15			
			Act means the Crimes (Sentencing Procedure) Amendment Options) Act 2017.	16 17			
	Offenders in Custody (Part 2A)						
	(1)	in custody includes a r before its su <b>Note</b> . Before	to a person described in paragraph (c) of the definition of <i>offender</i> or <i>offender</i> in section 26A as substituted by the amending Act reference to an offender described in that paragraph immediately abstitution.  It is substitution by the amending Act, paragraph (c) of that definition of offender within the meaning of Part 4 (Imprisonment by way of home	19 20 21 22 23 24			
		detention) of	the Crimes (Administration of Sentences) Act 1999.	25 25			
	(2)	includes a (Sentencing by the amen	to a community correction order in paragraph (e) of that definition reference to a community service order under the <i>Crimes Procedure</i> ) <i>Act 1999</i> as in force before the amendment of that Act ding Act. The community service order is taken to be a community order (see Part 29 of Schedule 2 to the <i>Crimes (Sentencing Act 1999)</i> .	26 27 28 29 30 31			
4.9	Crimes A	ct 1900 No	o 40	32			
	Section 31	OH Application	on of Part	33			
	Omit "or ho	me detention	n order" from section 310H (a).	34			
4.10	Crimes (A	Appeal and	d Review) Act 2001 No 120	35			
[1]	Section 3 [	efinitions		36			
	Insert after	paragraph (a)	(ii) of the definition of <i>sentence</i> :	37			
		(iia)	any community correction order, conditional release order or fine, and	38 39			

[2]	Sect	ion 3	(3) (c)			1		
	Inser	t at the	e end o	f section	on 3 (3) (b):	2		
				, and		3		
			(c)	cond	erence to varying or revoking a condition of, or imposing a new ition on, an intensive correction order, community correction order inditional release order.	4 5		
4.11	Crin	nes (	Forer	nsic F	Procedures) Act 2000 No 59	7		
	Section 3 Interpretation							
	Omit	the de	efinitio	n of <i>pl</i>	lace of detention from section 3 (1). Insert instead:	ę		
			order (Sent pursi	r that tencing uant to . A hom	tention, in relation to a person the subject of an intensive correction is subject to a home detention condition under the <i>Crimes a Procedure</i> ) Act 1999, means the place at which the person resides the order.  The detention order is taken to be an intensive correction order that is subject betention condition. See Part 29 of Schedule 2 to that Act.	10 11 12 13 14 15		
4.12	Crin No 7		Inters	state <sup>-</sup>	Transfer of Community Based Sentences) Act 2004	16 17		
	Section 7							
	Omit the section. Insert instead:							
	7	Wha	t is a '	'comm	nunity based sentence"?	20		
		(1)			ity based sentence is:	21		
		(-)	(a)		nis jurisdiction—any of the following:	22		
			(u)	(i)	an intensive correction order under section 7 of the <i>Crimes</i> (Sentencing Procedure) Act 1999 (including a home detention order that is taken to be an intensive correction order subject to a home detention condition under Part 29 of Schedule 2 to that Act),	23 24 25 26 27		
				(ii)	a community correction order under section 8 of the <i>Crimes</i> (Sentencing Procedure) Act 1999 (including a community service order or good behaviour bond that is taken to be a community correction order under Part 29 of Schedule 2 to that Act),	28 29 30 31 32		
				(iii)	a conditional release order under section 9 of the <i>Crimes</i> (Sentencing Procedure) Act 1999 (including a good behaviour bond that is taken to be a conditional release order under Part 29 of Schedule 2 to that Act),	33 34 35 36		
				(iv)	an order under section 12 (1) of the <i>Crimes (Sentencing Procedure) Act 1999</i> suspending a sentence, a good behaviour bond entered into in accordance with the order and the sentence of imprisonment imposed in relation to the order, where the order was made before the repeal of that section by the <i>Crimes (Sentencing Procedure) Amendment (Sentencing Options) Act 2017</i> ,	37 38 39 40 41 42 43		
				(v)	a sentence declared by the regulations to be a community based sentence, and	44 45		

		(b)	for an interstate jurisdiction—a sentence that is a community based sentence under the corresponding law of the jurisdiction.	1		
	(2)		he purposes of subsection (1) (a), the following are taken to be a single munity based sentence:	3		
		(a)	an order referred to in subsection (1) (a) (i) and the sentence of imprisonment in relation to which the order is made,	5		
		(b)	an order referred to in subsection (1) (a) (iv) and the good behaviour bond and the sentence of imprisonment imposed in relation to the order,	7 8		
		(c)	any combination of 2 or more community based sentences referred to in subsection (1) (a) that is declared by the regulations to be a single community based sentence.	9 10 11		
4.13	Criminal	Appe	eal Act 1912 No 16	12		
[1]	Section 2 D	Definit	ions	13		
	Omit "or he section 2 (1		etention order" from paragraph (a) (i) of the definition of Sentence in	14 15		
[2]	Section 2 (	1), def	inition of "Sentence"	16		
	Omit parag	raph (a	a) (ii). Insert instead:	17		
			(ii) any community correction order, conditional release order or community service order, and	18 19		
[3]	Section 2 (	1), def	inition of "Sentence"	20		
	Insert at the	end o	f the definition:	21		
		that c	Part 29 of Schedule 2 to the <i>Crimes (Sentencing Procedure) Act 1999</i> provides certain home detention orders, community service orders and good behaviour are taken to be intensive correction orders, community correction orders or tional release orders.	22 23 24 25		
4.14	Criminal	Proc	edure Act 1986 No 209	26		
	Chapter 7, Part 4 Intervention programs					
	Omit parag	raph (d	d) of the note that immediately follows the heading to the Part.	28		
4.15	Criminal	Reco	ords Act 1991 No 8	29		
[1]	Section 8 V	Vhen i	s a conviction spent?	30		
	Insert ", exc	cept as	provided by this section" after "made" in section 8 (2).	31		
[2]	Section 8 (	4) (c)		32		
	Insert at the	end o	f section 8 (4) (b):	33		
			or	34		
		(c)	the making of a conditional release order, without conviction, under section 9 of the <i>Crimes (Sentencing Procedure) Act 1999</i> , for a specified term and with 1 or more additional or further conditions imposed under that Act,	35 36 37 38		

4.16	Dangerous Go	oods	(Road and Rail Transport) Act 2008 No 95	1				
	Section 49 Operation of Division							
	Insert after section	n 49 (2	) (a):	3				
	(a1)	Crim	court makes a conditional release order under section 9 of the es (Sentencing Procedure) Act 1999 in respect of the offender in on to the offence (where the court does not proceed to conviction),	4 5 6 7				
4.17	Drug Court Ad	ct 199	98 No 150	8				
[1]			refer persons to Drug Court during proceedings for failure to s of good behaviour bond	9 10				
	Crimes (Sentencin	ng Pro	a court before whom an offender appears under section 98 of the <i>cedure) Act 1999</i> in connection with an alleged failure to comply good behaviour bond imposed under section 12 of that Act" from	11 12 13 14				
	Insert instead "It is the duty of a court before whom an offender appears under section 98 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as previously in force but continuing to apply under clause 76 (3) of Schedule 2 to that Act) in connection with an alleged failure to comply with the conditions of a good behaviour bond imposed under section 12 of that Act (as previously in force)".							
[2]	Section 7A Perso	ons ac	cepted into program in relation to offence	20				
	Omit section 7A (	(8) (b).	Insert instead:	21				
	(b)	be de	ealt with in proceedings under:	22				
		(i)	section 107C of the <i>Crimes (Administration of Sentences) Act</i> 1999 on the revocation of a community correction order, or	23 24				
		(ii)	section 108C of that Act on the revocation of a conditional release order, or	25 26				
		(iii)	section 98 or 99 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as previously in force but continuing to apply under clause 76 (3) of Schedule 2 to that Act) on the revocation of a good behaviour bond imposed under section 12 of that Act (as previously in force), whether or not the person has been referred to the Drug Court under section 7 in relation to the failure to comply with the conditions of the good behaviour bond, or	27 28 29 30 31 32 33				
[3]			cepted into program in relation to failure to comply with ity correction order, conditional release order or good	34 35 36				
	Omit section 7B (	1) (b).	Insert instead:	37				
	(b)	who	has been called on by the Drug Court to appear before it under:	38				
		(i)	section 107C of the <i>Crimes (Administration of Sentences) Act</i> 1999 in relation to a community correction order, or	39 40				
		(ii)	section 108C of that Act in relation to a conditional release order, or	41 42				
		(iii)	section 98 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as previously in force but continuing to apply under clause 76 (3) of Schedule 2 to that Act) in relation to a good behaviour bond under section 12 (as previously in force),	43 44 45 46				

[4]	Section 7B	3 (2)			1
	Insert "combond".	ımunit	y corre	ection order, conditional release order or" before "good behaviour	2
[5]	Section 7B	3 (2) (b	)		4
	Insert "orde	er or" l	efore '	"bond".	5
[6]	Section 7B	3 (3)			6
	Omit "good	l beha	viour b	ond by revoking the bond".	7
				ity correction order, conditional release order or good behaviour der or bond".	8 9
[7]	Section 7B	3 (4)			10
	Omit the su	bsecti	on. Ins	ert instead:	11
	(4)	The	provisi	ons of:	12
		(a)	not a	on 107D of the <i>Crimes (Administration of Sentences) Act 1999</i> do apply to a community correction order that is revoked under ection (3) of this section, and	13 14 15
		(b)		on 108D of that Act do not apply to a conditional release order that voked under that subsection, and	16 17
		(c)	previ Sche	on 99 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as ously in force but continuing to apply under clause 76 (3) of dule 2 to that Act) do not apply to a good behaviour bond that is seed under that subsection.	18 19 20 21
[8]	Section 7C with condi- behaviour	tions	ons all of com	owed to continue in program in relation to failure to comply munity correction order, conditional release order or good	22 23 24
	Omit sectio	n 7C (	(1) (b).	Insert instead:	25
		(b)	who	has been called on by the Drug Court to appear before it under:	26
			(i)	section 107C of the <i>Crimes (Administration of Sentences) Act</i> 1999 in relation to a community correction order, or	27 28
			(ii)	section 108C of that Act in relation to a conditional release order, or	29 30
			(iii)	section 98 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as previously in force but continuing to apply under clause 76 (3) of Schedule 2 to that Act) in relation to a good behaviour bond under section 12 (as previously in force),	31 32 33 34
[9]	Section 70	(3)			35
	Omit "good	l beha	viour b	ond by revoking the bond".	36
				ity correction order, conditional release order or good behaviour der or bond".	37 38

[10]	Sect	ion 7C	(4)			1		
	Omit	the su	ıbsecti	on. Ins	ert instead:	2		
		(4)	The	provisi	ons of:	3		
		(a)			on 107D of the <i>Crimes (Administration of Sentences) Act 1999</i> do apply to a community correction order that is revoked under ection (3) of this section, and	4 5 6		
			(b)	section is rev	on 108D of that Act do not apply to a conditional release order that voked under that subsection, and	7		
			(c)	previ Sche	on 99 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as lously in force but continuing to apply under clause 76 (3) of dule 2 to that Act) do not apply to a good behaviour bond that is ked under that subsection.	9 10 11 12		
[11]	Sect	ion 7D	) Pers	ons no	ot accepted into program in relation to offence	13		
	Omit	section	n 7D (	(8) (b).	Insert instead:	14		
			(b)	be de	ealt with in proceedings under:	15		
				(i)	section 107C of the <i>Crimes (Administration of Sentences) Act</i> 1999 on the revocation of a community correction order, or	16 17		
				(ii)	section 108C of that Act on the revocation of a conditional release order, or	18 19		
				(iii)	section 98 or 99 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as previously in force but continuing to apply under clause 76 (3) of Schedule 2 to that Act) on the revocation of a good behaviour bond imposed under section 12 of that Act (as previously in force), whether or not the person has been referred to the Drug Court under section 7 in relation to the failure to comply with the conditions of the good behaviour bond, or	20 21 22 23 24 25 26		
[12]	Sect	ion 7E				27		
	Omit the section. Insert instead:							
	7E	cond	Persons not accepted into program in relation to failure to comply with conditions of community correction order, conditional release order or good behaviour bond					
		(1)	This	section	n applies to a person:	32		
			(a)	who	has been referred to the Drug Court under section 7, or	33		
			(b)	relev	has been called on by the Drug Court to appear before it under the ant provisions referred to in subsection (3),	34 35		
			but v	whom t	he Drug Court has not dealt with under section 7B or 7C.	36		
		(2)			Court is to deal with a person to whom this section applies in with those relevant provisions.	37 38		
		(3)	For t	he pur	poses of this section, the relevant provisions are:	39		
			(a)		on 107C of the <i>Crimes (Administration of Sentences) Act 1999</i> in on to a community correction order, or	40 41		
			(b)	section	on 108C of that Act in relation to a conditional release order, or	42		
			(c)	previ Sche	on 98 of the <i>Crimes (Sentencing Procedure) Act 1999</i> (as lously in force but continuing to apply under clause 76 (3) of dule 2 to that Act) in relation to a good behaviour bond under on 12 (as previously in force).	43 44 45 46		

[13]	Section 12	Impos	ition of final sentence	1			
	Insert after	section	. 12 (4):	2			
	(5)	senter subsec (Sente Amen	the purpose of making an order under subsection (3) in relation to an initial nace imposed before the commencement of this subsection, ctions (3) and (4) have effect as if the amendments made to the Crimes tencing Procedure) Act 1999 by the Crimes (Sentencing Procedure) adment (Sentencing Options) Act 2017 were in force when the initial nace was imposed.	3 4 5 6 7 8			
[14]	Section 24	Jurisd	liction of Drug Court	9			
	Omit "sect section 24		(1) (b) and (c) of the Crimes (Sentencing Procedure) Act 1999" from	10 11			
	(Administro (Sentencing	ation of g Proce	tion 107C (1) (b) and (c) and section 108C (1) (b) and (c) of the <i>Crimes f Sentences</i> ) Act 1999 and section 98 (1) (b) and (c) of the <i>Crimes edure</i> ) Act 1999 (as previously in force but continuing to apply under chedule 2 to that Act)".	12 13 14 15			
4.18	Education	n Act	1990 No 8	16			
	Section 23	Offend	ce if parent fails to send child to school	17			
	Omit "community service order under section 8 (1) of the <i>Crimes (Sentencing Procedure) Act 1999</i> " from section 23 (5).						
	Procedure)	Act 19	mmunity correction order under section 8 of the <i>Crimes (Sentencing 199)</i> that is subject to the standard conditions of a community correction amunity service work condition".	20 21 22			
4.19	Fines Ac	t 1996	6 No 99	23			
[1]	Section 3 Definitions						
	Insert in al	ohabetic	cal order in section 3 (1):	25			
	,		nunity service order—see section 79.	26			
	Note. A community service order applies in relation to a child.						
[2]	Section 58	Summ	nary of enforcement procedure	28			
	Omit section	on 58 (1	) (d)–(f). Insert instead:	29			
		(d)	Order requiring community service	30			
		, ,	If civil enforcement action is not successful, an order requiring community service is served on the fine defaulter (see Division 5).	31 32			
		(e)	Imprisonment if failure to comply with order	33			
			If the fine defaulter does not comply with the order requiring community service, a warrant of commitment is issued to a police officer for the imprisonment of the fine defaulter (except in the case of children).	34 35 36 37			
		(f)	Fines payable by corporations	38			
			The procedures for fine enforcement (other than order requiring community service work and imprisonment) apply to fines payable by corporations (see Division 7).	39 40 41			

[3]	Part	4, Divi	sion 5	1					
	Omi	t "Com	munity service orders" from the heading.	2					
	Inser	t instea	ad "Orders requiring community service".	3					
[4]	Sect	ion 79		4					
	Omit the section. Insert instead:								
	79	79 Making of order against fine defaulter							
		(1)	The Commissioner may make an order under this Division requiring a fine defaulter to perform community service work in order to work off the amount of the fine that remains unpaid.	7 8 9					
		(2)	An order may be either:	10					
			(a) in the case of an adult—a community correction order subject to a community service work condition, or	11 12					
			(b) in the case of a child—a community service order.	13					
		(3)	The Commissioner may make an order only if satisfied that enforcement action is authorised against the fine defaulter under this Division.	14 15					
		(4)	The Commissioner is not to make an order:	16					
			(a) if satisfied that the person is not capable of performing work under an order or is otherwise not suitable to be engaged in such work, and	17 18					
			(b) unless the Commissioner has obtained a report from an appropriate officer concerning the person's suitability to be so engaged and the report indicates that the person is assessed as suitable.	19 20 21					
		(5)	An order may be made in the absence of, and without notice to, the fine defaulter.	22 23					
		(6)	An order may be made before the execution of a warrant under Division 4 and in anticipation that enforcement action may not be successful under that Division in satisfying the fine. The order is not to be served if that enforcement action is successful.	24 25 26 27					
		(7)	Subject to this Division, such of the provisions of the <i>Crimes (Sentencing Procedure) Act 1999</i> , the <i>Crimes (Administration of Sentences) Act 1999</i> and the regulations under those Acts as are prescribed for the purposes of this subsection by the regulations under this Act:	28 29 30 31					
			(a) apply to a community correction order under this Division in the same way as they apply to a community correction order under the <i>Crimes</i> (Sentencing Procedure) Act 1999, and	32 33 34					
			(b) so apply with any modifications so prescribed.	35					
		(8)	Subject to this Division, section 11 and such other provisions of the <i>Children</i> (Community Service Orders) Act 1987 and the regulations under that Act:	36 37					
			(a) apply to a community service order under this Division in the same way as they apply to a community service order under that Act, and	38 39					
			(b) so apply with any modifications so prescribed.	40					
			<b>Note.</b> Section 11 of the <i>Children (Community Service Orders) Act 1987</i> requires a community service order to require the offender to perform graffiti clean up.	41 42					
		(9)	An order under this section is to be in the approved form.	43					

	(10)	For the purposes of this section, an appropriate officer is:	1				
		(a) in the case of an adult—a community corrections officer (as defined in the <i>Crimes (Administration of Sentences) Act 1999</i> ), or	2				
		(b) in the case of a child—a juvenile justice officer (as defined in the <i>Children (Detention Centres) Act 1987</i> ).	4 5				
[5]	Sections 8	0, 80A, 81, 82, 83, 84, 85, 86, 87 and 88	6				
		mmunity service order", "A community service order", "the community service "The community service order" wherever occurring.	7 8				
	Insert instead	ad "an order", "An order", "the order" and "The order" respectively.	9				
[6]	Sections 8	1 and 86	10				
	Omit "othe occurring.	er community service order" and "other community service orders" wherever	11 12				
	Insert instead	ad "other such order" and "other such orders" respectively.	13				
[7]	Section 87	Imprisonment following breach of order	14				
	Omit "com	munity service" from section 87 (2).	15				
[8]	Section 87	Imprisonment following breach of order	16				
	Omit "com	munity service orders" from section 87 (2). Insert instead "orders".	17				
[9]	Section 89	Intensive correction order	18				
	Omit section 89 (1) and (2). Insert instead:						
	(1)	If a fine defaulter is committed to a correctional centre by warrant under this Division, the Commissioner of Corrective Services may determine that the fine defaulter's period of imprisonment be served under an intensive correction order. The determination may be made in anticipation of the issue of a warrant.	20 21 22 23 24				
	(2)	The Commissioner of Corrective Services may, having made a determination under subsection (1), make an intensive correction order that is subject to:	25 26				
		(a) the standard conditions of an intensive correction order under the <i>Crimes (Sentencing Procedure) Act 1999</i> , and	27 28				
		(b) if the Commissioner of Corrective Services so elects—a community service work condition under that Act.	29 30				
[10]	Section 89	(5)	31				
	Omit the su	absection. Insert instead:	32				
	(5)	The decision whether or not to make a determination under this section is at the discretion of the Commissioner of Corrective Services, and no appeal lies against a decision, determination or order under this section in a particular case.	33 34 35 36				
[11]	Section 89	(8)	37				
	Omit the su	absection. Insert instead:	38				
	(8)	Subject to this Division, such of the provisions of the <i>Crimes (Sentencing Procedure) Act 1999</i> , the <i>Crimes (Administration of Sentences) Act 1999</i> and the regulations under those Acts as are prescribed for the purposes of this subsection by the regulations under this Act:	39 40 41 42				

		(a)	apply to an intensive correction order under this Division in the same way as they apply to an intensive correction order under the <i>Crimes</i> (Sentencing Procedure) Act 1999, and	2	
		(b)	so apply with any modifications so prescribed.	4	
[12]	Section 98	Appli	ication of civil and other enforcement procedures	Ę	
	Omit "Con	munit	sy service orders". Insert instead "Orders requiring community service".	6	
[13]	Section 99	B Mak	king an order	7	
	Omit "a co	mmun	ity service order is not in force" from section 99B (1) (c).	8	
	Insert instead "a community correction order or community service order is not in force under Division 5".				
[14]	Sections 1	00 an	d 101	11	
	Omit "a co	mmun	ity service order is issued" wherever occurring.	12	
	Insert inste Division 5'		community correction order or community service order is issued under	13 14	
[15]	Section 10	3 Elec	ctronic transmission of documents	15	
	Omit "a co	mmun	ity service order" from section 103 (1) (c).	16	
	Insert inste	ad "a c	community correction order or community service order".	17	
[16]	Section 12	0 Gui	delines on exercise of functions under this Act	18	
	Omit "or co	ommui	nity service orders" from section 120 (1) (a).	19	
	Insert inste under Divis		the issue of community correction orders or community service orders of Part 4".	20 21	
[17]	Section 125 Abolition of imprisonment as primary enforcement action for fine default				
	Omit "a co	mmun	ity service order" from section 125 (2).	23	
	Insert inste Division 5	ad "a of Par	community correction order or community service order made under t 4 and".	24 25	
[18]	Schedule 3	3 Savi	ngs, transitional and other provisions	26	
	Insert at the	e end c	of the Schedule, with appropriate Part and clause numbering:	27	
	Part	Pro	ovisions consequent on enactment of Crimes	28	
		(Se	entencing Procedure) Amendment (Sentencing	29	
		Op	tions) Act 2017	30	
	Division	1	Preliminary	3′	
	Defi	nitions	5	32	
			is Part:	33	
			t means a person who is of or above the age of 18 years.	34	
		(Sen	nding Act means the Crimes (Sentencing Procedure) Amendment tencing Options) Act 2017.	35 36	
		chila	<i>I</i> means a person who is under the age of 18 years.	37	

Division 2	2 Existing orders	1
Exist	ing community service orders under section 79 for adults	2
(1)	This clause applies to a community service order made under section 79 before the date of the substitution of that section by the amending Act (the <i>substitution date</i> ), being an order that was:	3 4 5
	(a) made in respect of an adult, and	6
	(b) in force immediately before the substitution date.	7
(2)	The community service order is taken to be a community correction order made under section 79 as substituted by the amending Act.	8 9
(3)	Without limiting subclause (2), the community correction order is subject to a community service work condition referred to in section 79 as substituted by the amending Act.	10 11 12
(4)	The conditions applicable under this Act as in force before the substitution date in respect of the community service order do not apply to the community correction order.	13 14 15
Exist	ing community service orders under section 79 for children	16
(1)	This clause applies to a community service order made under section 79 before the date of the substitution of that section by the amending Act (the <i>substitution date</i> ), being an order that was:	17 18 19
	(a) made in respect of a child, and	20
	(b) in force immediately before the substitution date.	21
(2)	The community service order is taken to be a community service order made under section 79 as substituted by the amending Act.	22 23
(3)	The community service order is subject to the conditions prescribed by or determined under the regulations for the purposes of this clause.	24 25
(4)	The conditions applicable under this Act as in force before the substitution date in respect of the order do not apply to the community service order.	26 27
Exist	ing intensive correction orders under section 89	28
(1)	This clause applies to an intensive correction order made under section 89 (2) before the date of the substitution of that subsection by the amending Act (the <i>substitution date</i> ), being an order that was in force immediately before the substitution date.	29 30 31 32
(2)	The intensive correction order is taken to be an intensive correction order made under section 89 (2) as substituted by the amending Act.	33 34
(3)	Without limiting subclause (2), the intensive correction order is subject to a community service work condition referred to in section 89 (2) as substituted by the amending Act.	35 36 37
(4)	The conditions applicable under this Act as in force before the substitution	38

date in respect of the order do not apply to the intensive correction order.

39

4.20	<b>Fisheries</b>	Management Act 1994 No 38	1
[1]	Sections 22	20ZGA Community service work	2
	Omit "a con	nmunity service order" from section 220ZGA (1).	3
		d "an intensive correction order or community correction order respectively community service work condition".	4 5
[2]	Section 220	)ZGA (3)	6
	Omit the det	finition of community service order.	7
[3]	Section 220	)ZGA (3)	8
	Insert after t	he definition of <i>community service work</i> :  intensive correction order and community correction order have the same meanings as they have in the Crimes (Sentencing Procedure) Act 1999.	9 10 11
[4]	Section 282	2K Community service work in respect of serious fisheries offences	12
	Omit "a con	nmunity service order" from section 282K (1).	13
		d "an intensive correction order or community correction order respectively community service work condition".	14 15
[5]	Section 282	2K (3)	16
	Omit the det	finition of <i>community service order</i> .	17
[6]	Section 220K (3)		18
	Insert after t	he definition of community service work:	19
		intensive correction order and community correction order have the same meanings as they have in the Crimes (Sentencing Procedure) Act 1999.	20 21
4.21	Graffiti Co	ontrol Act 2008 No 100	22
[1]	Section 9Q Application of sentencing legislation to orders made in respect of adult offenders		
	Omit "comn	nunity service order" from section 9Q (1).	25
	Insert instead "community correction order".		26
[2]	Section 9Q	(2)	27
	Omit the sub	osection. Insert instead:	28
	(2)	Such of the provisions of the <i>Crimes (Sentencing Procedure) Act 1999</i> or the <i>Crimes (Administration of Sentences) Act 1999</i> as may be prescribed by the regulations under this Act do not apply to a community clean up order.	29 30 31
[3]	Section 13B Alternative and additional actions to imposing penalty for graffiti offences under sections 4 (2) and 5		32 33
	Omit section 13B (1). Insert instead:		
	(1)	A court may, instead of imposing a fine on a person for an offence under section 4 (2) or 5, make:	35 36
		(a) a community correction order under the <i>Crimes (Sentencing Procedure)</i> Act 1999 that is subject to the standard conditions of a community correction order and to a community service work condition, or	37 38 39

	(b) a community service order under the <i>Children (Community Service Orders) Act 1987</i> directing the person to perform community service work,	1 2 3		
	as the case requires.	4		
4.22	Jury Act 1977 No 18	5		
[1]	Schedule 1 Persons excluded from jury service	6		
	Insert ", or an intensive correction order that is subject to a home detention condition," after "home detention order" in clause 2 (5) (a).	7		
[2]	Schedule 1, clause 4	9		
	Omit "or an order under section 9, 10 or 11" from clause 4 (1) (b).	10		
	Insert instead ", a community correction order, a conditional release order or an order under section 10 or 11".	11 12		
4.23	Motor Dealers and Repairers Regulation 2014	13		
	Clause 37 Certain persons not to be employed as manager	14		
	Insert after clause 37 (1) (b):	15		
	(b1) the person is subject to a community correction order or conditional release order in relation to an offence involving fraud or dishonesty, or	16 17		
4.24	Privacy Code of Practice (General) 2003	18		
	Clause 12 Definitions	19		
	Omit paragraph (d) of the definition of <i>offender</i> in clause 12 (1). Insert instead:			
	(d) a person who is the subject of an intensive correction order, a community correction order, a conditional release order (with or without conviction), an intervention program order, a non-association order or a place restriction order within the meaning of the <i>Crimes</i> (Sentencing Procedure) Act 1999 that is in force,	21 22 23 24 25		
4.25	Security Industry Regulation 2016	26		
[1]	Clause 15 Offences and civil penalties that disqualify applicants	27		
	Insert "an intensive correction order, a community correction order, a conditional release order," after "(whether or not suspended)," wherever occurring in clause 15 (1) (b) and (c).	28 29		
[2]	Clause 15 (1) (c) (i)			
	Omit ", or both".			

4.26	<b>Summary Offe</b>	ences Act 1988 No 25	1
[1]	Section 4A Offer	nsive language	2
	Omit section 4A (3) (a). Insert instead:		
	(a)	may make a community correction order under section 8 of the <i>Crimes</i> (Sentencing Procedure) Act 1999 that is subject to the standard conditions of a community correction order and to a community service work condition (despite the offence not being punishable by imprisonment), or	4 5 6 7 8
[2]	Section 8 Damag	ging or desecrating protected places	9
	Omit section 8 (3A) (a). Insert instead:		
	(a)	may make a community correction order under section 8 of the <i>Crimes</i> (Sentencing Procedure) Act 1999 that is subject to the standard conditions of a community correction order and to a community service work condition (despite the offence not being punishable by imprisonment), or	11 12 13 14 15
4.27	Supreme Cou	rt Act 1970 No 52	16
	Section 69C Stay	y of execution of conviction, order or sentence pending review	17
	Omit ", or home of	detention order," from section 69C (6).	18
4.28	Tow Truck Inc	dustry Regulation 2008	19
	Clause 7 Offences that disqualify applicants for licences		
	Insert "an intensive correction order, a community correction order under section 8 (1) of the <i>Crimes (Sentencing Procedure) Act 1999</i> that is subject to a condition requiring the person to perform community service work for 100 hours or more," after "imprisonment," in clause 7 (1).		